The Consequence of the Servant's Suffering for the Relationship between God and the Others in Isaiah 53

Master's Thesis in Theology NLA University College Bergen Fall 2017 Joanna Bauer

Errata

Page	Line	Correction
17	8 from bottom	transpose מִשְׁחָת and מִשְׁחָת
18	2 from top	read מִשְׁחָת for מִשְּׁחָת
31	2 from top	transpose מִשְׁחַת and מִשְׁחָת
	3 from top	transpose מִשְׁחַת and מִשְׁחָת
43	12 from bottom	transpose מִשְׁחָה and מִשְׁחָה

Abstract

This thesis studies the relation between the servant's suffering and the others' reconciliation with God as it is presented in Isaiah 53.

Commonly, reconciliation in Isaiah 53 is suggested to be by the servant's message causing contrition or faith, or by his suffering being atoning. Reconciliation by message is commonly based on interpreting the servant as a prophet. Reconciliation by atonement is commonly based on expressions in Isaiah 53 which can be related to cultic expiation of sin.

The thesis is based on a thorough study of the text of Isaiah 53. This study suggests uncommon answers to textcritical questions. It also suggests uncommon translations of some verses of Isaiah 53. Further, it suggests that Isaiah 53 is intentionally ambiguous, both concerning the identity of the servant, and concerning reconciliation.

Isaiah 53 depicts the servant both as in need of reconciliation and as righteous, and thus not in need of reconciliation. This contradiction can be resolved by an ambiguous identity of the servant. Isaiah 53 depicts the servant ambiguously both as the servant Israel, in need of reconciliation, and as a righteous servant, who reconciles Israel.

Isaiah 53 states reconciliation by chastisement and insight, and by transfer of the burden of sin. Isaiah 53 also alludes to cultic expiation of sin. Concerning chastisement, the servant Israel is chastised by being sent into exile. This chastisement fosters the servant's insight and resulting righteous conduct, and thereby, the servant's peace with God. Concerning transfer of the burden of sin, Isaiah 53 states that the we-group's and the many's burden of sin, that is, the guilt incurred by sin and the punishment for this guilt, is transferred to the righteous servant. By this transfer, the others are freed from their guilt and thus reconciled. Concerning cultic expiation of sin, Isaiah 53 clearly alludes to both the Day of Atonement and the liability offering, but does not decisively depict the servant as scapegoat or liability offering.

Reconciliation by transfer of the burden of sin amounts to vicarious suffering. The concept of vicarious suffering has been criticized for being unique to the Old Testament. However, vicarious substitution is a concept commonly found in the Old Testament. Vicarious substitution may have inspired the concept of vicarious suffering as a specific application.

Forord

Takk til veilederen min, Kåre Berge, for å utvikle meg. Hos deg fikk jeg det som jeg ønsket meg mest: utfordringer. Takk for alle samtalene, som jeg har lært mye av, og som også førte til idéen for masteroppgaven.

Takk til både Kåre og biveilederen min, Kristin Joachimsen, for mange kritiske kommentarer og konstruktive tilbakemeldinger til oppgaven. Takk for at jeg ble utfordret på tjenerens identitet og på ordstudiene. Det førte til en bedre forståelse av Jesaja 53, som var det jeg ønsket meg.

Takk til Gunnar Innerdal for konstruktiv kritikk. Også den førte til bedre forståelse av teksten.

Takk til NLA for et bra studiemiljø, og for finansiell støtte til biveiledning.

Takk til bibliotekarene på NLA, Marianne Grøvlen Sandven og Gunn Fasmer, for stor tålmodighet for å skaffe tyske Jesajakommentarer.

Takk til medstudenter og venner for støtte i form av både oppmuntring og avkobling. Herzlichen Dank an meine Familie und Freunde für Eure Unterstützung.

Danke an meinen Herrn, Jesus Christus, der sich mit mir verbündet hat und die Last meiner Sünden trug.

Joanna Bauer bauerj@fs.tum.de

Contents

1	Intr	oducti	ion	9
	1.1	Previo	ous Work	11
		1.1.1	Reconciliation by Message	11
		1.1.2	Reconciliation by Atonement	12
	1.2	The Io	dentity of the Servant and the Context of Isaiah 53	13
	1.3	Concl	usion	16
2	$\mathrm{Th}\epsilon$	Text	of Isaiah 53	17
	2.1	Differe	ences between MT and 1QIs ^a	17
	2.2	Expia	tion of Sin versus Punishment of Sin	19
		2.2.1	The Burden of Sin	19
		2.2.2	Expressions Related to Expiation of Sin	20
		2.2.3	Expressions Related to Judgment and Punishment of Sin	24
		2.2.4	Conclusion	26
	2.3	Expre	ssions Related to Knowing God and Keeping His Commandments	27
	2.4	Transl	ation	28
		2.4.1	Comments	30
	2.5	The T	Text of the LXX	36
	2.6	Struct	sure	38
3	$\mathrm{Th}\epsilon$	Ambi	iguous Identity of the Servant	43
	3.1	Ambig	guities in Isaiah 53	43
	3.2	The A	Embiguous Identity of the Servant	44
4	Sufl	fering i	in Isaiah 53	47
	4.1	Inclus	ive or Exclusive Suffering	48
		4.1.1	Objections against the Servant Suffering Exclusively	49
		4.1.2	The Servant's Exclusion	50
	4.2	Concl	usion	51

5	The	Cons	equence of the Servant's Suffering	53
	5.1	Comm	non Interpretations of the Servant's Reconciling Ministry	53
		5.1.1	Reconciliation by Message	53
		5.1.2	Reconciliation by Atonement	55
	5.2	Recon	ciliation by Transfer of the Burden of Sin	57
		5.2.1	Objections Against Transfer of the Burden of Sin	58
	5.3	Recon	ciliation by Chastisement and Insight	60
	5.4	Concl	usion	61
6	Vica	arious	Substitution in the Old Testament	65
	6.1	Concl	asion	67
7	Con	clusio	ns	69
Bi	bliog	graphy		73

Abbreviations

$1 \mathrm{QIs^a}$	The almost complete Isaiah scroll from Qumran's cave 1
$1 \mathrm{QIs^b}$	The other Isaiah scroll from Qumran's cave 1
$4 \mathrm{QIs^d}$	Isaiah fragments containing 53:8–12 from Qumran's cave 4
LXX	The Septuagint
МТ	The Masoretic Text
NRSV	The New Revised Standard Version

Chapter 1

Introduction

Isaiah 52:13-53:12 (for convenience: Isaiah 53) is commonly understood to present an extraordinary constellation. The constellation consists of, on the one hand, groups designated only as "we" and "the many," clearly depicted as sinful, and, on the other hand, "my servant," depicted as suffering and righteous. The extraordinary aspect of this constellation, as it is commonly understood, is that the righteous servant's suffering reconciles the sinful groups with God. However, there is no common understanding on how the servant's suffering is related to reconciliation. This relation between the servant's suffering and reconciliation is studied in this thesis.

The study starts with reviewing recent influential interpretations of Isaiah 53. These suggest two interpretations of the relation between the servant's suffering and reconciliation. One of these depends strongly on the supposed identity of the servant, which in turn depends on the supposed context of Isaiah 53. Consequently, studying the relation between the servant's suffering and reconciliation requires studying both the identity of the servant, and the context of Isaiah 53. This thesis studies the context of Isaiah 53, the identity of the servant, and the relation between the servant's suffering and reconciliation as follows:

Section 1.1 reviews five recent influential interpretations of Isaiah 53. They disagree both on the context of Isaiah 53, and the identity of the servant, and the relation between the servant's suffering and reconciliation. However, the reviewed interpretations exhibit a pattern. The relation between suffering and reconciliation is interpreted to be either by prophetic message or by atoning suffering. The former always coincides with identifying the servant as the prophet Deuteroisaiah, and the so-called servant songs as the context. On the contrary, the latter always coincides with identifying the servant as Israel, and supposing Isaiah 40–55 as the context of Isaiah 53. It seems that whether reconciliation is interpreted to be by atonement or message, is strongly dependent on supposed identity, which in its turn is dependent on the supposed context.

Consequently, Section 1.2 briefly discusses the context of Isaiah 53 and the identity of the servant. Both the question of the context of Isaiah 53 and of the identity of the servant

are debated. Answering them requires a detailed study of both language, structure, and content of Isaiah 40–55, which cannot be undertaken within a master's thesis. However, since the arguments for excising the servant songs do not seem to be compelling, it seems reasonable to assume that Isaiah 40–55 is the context of Isaiah 53, and consequently, that the servant is Israel.

All reviewed interpretations that consider Isaiah 40-55 to be the context of Isaiah 53, interpret the servant's suffering as atoning. This is motivated by expressions in Isaiah 53 which can be related to cultic expiation of sin. Consequently, the next step is to study these. In order to do so, Section 2.1 together with Section 2.5 establish the textual basis of Isaiah 53. Sections 2.2.1 and 2.2.2 study expressions in Isaiah 53 which can be related to cultic expiation of sin. Since sin alternatively can be condemned and punished, Section 2.2.3 studies expressions in Isaiah 53 which can be related to judgment and punishment of sin. Section 2.3 studies expressions related to avoiding to sin, that is, to knowing God and keeping his commandments. Of these three groups of expressions, those related to judgment and punishment of sin seem to constitute the majority. Section 2.4 presents a corresponding translation. Section 2.6 studies the structure of Isaiah 53.

The study of the text of Isaiah 53 shows that the rhetoric device of ambiguity is employed frequently. Section 3.1 lists these ambiguities. The frequency of the rhetoric device of ambiguity suggests that it is employed intentionally, in order to create an ambiguous text. Specifically, verse $11a\gamma$ has several meanings which seem to contradict each other, with some stating that the servant is in need of reconciliation, while others state that the servant is righteous, and thus not in need of reconciliation. Since the ambiguity seems to be intended, this suggests that the identity of the servant is ambiguous. This is discussed in Section 3.2, suggesting that the servant seems to ambiguously designate both the servant Israel in need of reconciliation with God, and a righteous servant, who reconciles the servant Israel.

Chapter 5 studies the relation between the righteous servant's suffering and reconciliation. Section 5.1.1 discusses the reviewed suggestion of reconciliation by the servant's message fostering contrition or faith. This interpretation mixes up elements of reconciliation by transfer of the burden of sin and of reconciliation by chastisement, discussed in Sections 5.2 and 5.3, respectively. Section 5.1.2 discusses the reviewed suggestions of the servant's suffering being atoning. Isaiah 53 alludes to both atoning sacrifices and to the Day of Atonement, but the servant is not decisively depicted as an atoning sacrifice or the scapegoat. Section 5.2 suggests that reconciliation is achieved by transfer of guilt for sin and punishment of this guilt to the servant. Section 5.2.1 discusses potential objections against this suggestion. Section 5.3 suggests that reconciliation also happens by chastisement, fostering appropriate fear of God and corresponding righteous conduct.

Transfer of punishment is traditionally called *vicarious suffering*. The servant suffering as a substitute requires that the servant suffers instead of those he substitutes for. This

in turn requires that the servant suffers alone or at least distinctively more profoundly than the others. Since this is a prerequisite, Chapter 4 studies whether the servant suffers alone or along with others. The descriptions of suffering in Isaiah 53 suggest that both is the case, with the focus of Isaiah 53 on the servant's exclusive suffering.

Since the concept of vicarious suffering commonly is criticized for being unique to the Old Testament, Chapter 6 suggests Old Testament texts which treat vicarious substitution, and thus may have inspired the concept of vicarious suffering.

Chapter 7 summarizes the results.

1.1 Previous Work

To the best of my knowledge, there does not exist an overview over proposals on how the servant's ministry is related to reconciliation. In the following, recent influential interpretations of this relation are reviewed. They present two different conceptions of this relation, reconciliation by prophetic message, or by atonement.

1.1.1 Reconciliation by Message

Both Janowski (1993, 8) and Hermisson (2017b, 417) accept the hypothesis of Duhm (1892, 14), who considers Isaiah 53 together with Isaiah 42:1–4, 49:1–6, and 50:4–9 to constitute four *servant songs* originally independent of Isaiah 40–55. Thus, both Janowski (1993, 8) and Hermisson (2017b, 417) consider the first three servant songs to be the original context of Isaiah 53. Both consider the servant of these songs to be the prophet Deuteroisaiah (Janowski 1993, 10; Hermisson 2017b, 417). They seem to interpret reconciliation as comprising two stages. As a prerequisite for reconciliation, the servant dies as compensation for Israel's guilt. Reconciliation ultimately happens by contrition, fostered by the servant's message, or by faith into the servant's message:

Israel to compensate for its obligation. Thus, in order to have a future, Israel had to be released from its obligation. This happens by the servant dying as 章以, "Schuldtilgung." The servant's death together with the message of Is 52:13–53:1 causes Israel to understand the meaning of the servant's death, and by this, its own situation of being guilty (13, 20). This insight of being guilty causes Israel to confess its guilt in Is 53:4–6. Only this contrite Israel is reconciled with God (20) (and identically: Janowski (1997, 90, 82, 91f)). Thus, the servant dying as "Schuldtilgung" constitutes only a prerequisite to reconciliation, which ultimately happens by insight and contrition, fostered by the prophet's message together with his death.

Similarly, Hermisson (2017b, 418) interprets Israel as guilty of not believing the servant's message. But instead of making the people endure the burden of their guilt, God

diverts it on the servant (Hermisson 2017b, 418), who dies as אַשָּׁשְּ, "vollgültige Sühneleistung," a merit which fully compensates for the guilt of the servant's people (398). His resurrection causes Israel to believe in the servant's message of JHWH's plan for salvation. Since their guilt was their unbelief in his message, and they now believe it, they are reconciled with God (424).

1.1.2 Reconciliation by Atonement

Alternatively, the servant's suffering or death is interpreted as cultic expiation of sin. There are different understandings of how exactly the servant's suffering is atoning. In the following, three of them are presented.

The Servant as Scapegoat

Mettinger (1983) sets out by doubting Duhm's hypothesis of four originally independent servant songs. Mettinger (1983, 18–22) suggests that the songs are an integral part of the structure of Isaiah 40–55, and thus must have been part of the original composition. Thus, the servant of the servant songs must be the same as the servant in the other servant passages of Isaiah 40–55, whom Mettinger (1983, 43) identifies as the Israelites in Babylonian exile.

Mettinger (1983, 41) suggests that the exiled community achieves reconciliation by acting as Israel's scapegoat. He substantiates this as follows: First, אָרֶץ הַיִּים should be interpreted as the opposite of desert. Thus, Is 53:8 describes the servant as being driven into the desert, like the scapegoat on the Day of Atonement. Second, the scapegoat's carrying away of the sins is expressed by נשא שון, which Mettinger (1983, 41), following Zimmerli (1969, 236–244), finds alluded to by the expressions סבל מון (Is 53:4), מלא המאם (Is 53:1), and נשא המא (Is 53:12).

The Servant as Cultmetaphorical Compensation

Berges (2008, 42) assumes that Isaiah 40–55 was written starting in 550 BC by a group closely connected to levitic temple musicians, and that Isaiah 40–55 was connected to Isaiah 1–39 when this group returned to Judah. The servant is interpreted as a fictional figure, a personification of the "Prototyp derer, die ab Jes 54,17b 'Knechte' genannt [werden]," who in turn are those who identify themselves with this fictional servant (Berges 2015, 229f).

Berges (2015, 269) interprets אָשָׁלָּא, and thereby the servant, based on 1.Sam 5f as a "cultmetaphorical" compensation as follows: JHWH diverts "die Exilsschuld des ganzen Volkes" on the servant and consequently strikes the servant with sickness (269). This is legitimate because the servant represents Israel (256). The people first disregard the servant as struck by God, but then realize that since it was for their sins that he was

struck, it was unjust to disregard him. Reconciliation with God and each other is possible, if the people then acknowledge his suffering as their אָשָׁבּ (269): Just as the Philistines in 1.Sam 5f sent a representation of their suffering as אַשָּׁב, so the Israelites can set a representation of their suffering (being exiled and disregarded), that is, their suffering and disregarded representative, as theirs.

The Servant as Guilt Offering

Sweeney (2016, 29f, 33) regards Isaiah 40–55 as originally written by the supposed prophet Deuteroisaiah, and having undergone redaction until the mid-fifth to early forth century BC. The servant is interpreted as all Israelites who have suffered in the period between the Assyrian invasion in the eighth century up to the final redaction of the book of Isaiah in the mid-fifth to early forth century BC (215).

Sweeney (2016, 216) interprets the people's relationship to God as marred by two issues. Regarding the first, the people's sin, reconciliation is achieved by the servant being "led to slaughter like a lamb" (Is 53:7), that is, sacrificed as an שַּשָּׁא, "guilt offering" (213, 216).

The second issue is less relevant for the following, since it might be characterized as reconciliation of God with the people rather than the other way round. Due to the people's experience of defeat and exile, they doubt that JHWH is still on their side, or that he has the power to protect them. Concerning this issue, God is reconciled with the people by restoring Israel (215).

1.2 The Identity of the Servant and the Context of Isaiah 53

Interpreting the reconciling ministry of the servant to be by his prophetic message seems to be closely connected to identifying the servant with a prophet, as both Janowski (1993) and Hermisson (2017b) do. This is substantiated by that those who interpret the servant as achieving reconciliation by his atoning suffering, do not identify the servant with a prophet. Consequently, this section briefly studies the identity of the servant. As will be seen, the supposed identity of the servant is closely related to the supposed context of Isaiah 53.

עבד occurs 21 times in Isaiah 40–55. In 54:17, it is used in plural, and in 44:26, it is paralleled with מַלְאָבִין, "his messengers." מַלְאָבִין is used together with both "Jacob" and "Israel" in 41:8, 41:9, twice in 42:19 (together with 42:24), 43:10 (together with 43:1 and 43:22), 44:1, 44:2 (with "Jeshurun" for "Israel"), twice in 44:21, and 45:4. In 48:20, it is used only with "Jacob," but the context "Go out from Babel!" indicates that Israel is meant. Within the servant songs, in 49:3, "used in combination with "Israel", and

in 42:1, 49:7, 50:10, 52:13, and 53:11, without a name connected to it. The two remaining occurrences in 49:5 and 49:6 are discussed below.

Identifying the servant with Israel is commonly based on the following observations: First, עבד in the majority of occurrences designates Israel. Second, שבד explicitly designates Israel in its first occurrence, 41:8. Thus, it can be argued that later occurrences do not need to repeat the designation, but can presuppose that it is already known that the servant is Israel. Finally, the servant of the servant songs is described by the same attributes as the servant Jacob/Israel:

```
chosen (servant Jacob/Israel: בְּחַרְמִּיךְ 41:8f; servant of the songs: בְּחַירִי 42:1)
called (קְרָאִנִי 41:9; קְרָאִנִי 49:1)
upheld (קְרָאִנִי 41:10; אָחְמֶּדְ־בּוֹ 42:1)
formed in the womb (נצרי מָבֶּטֶן 44:2,24; יצרי מָבֶּטֶן 49:5)
honored in the sight of JHWH (בְעֵינֵי נְכְבַּרְהָּ 43:4; אָפָבֵר בְעֵינֵי 43:4; אָפָבֵר בְעֵינֵי 49:5)
JHWH will be glorified in him (בְּמַלָּרְאָל יִחְפָּאָר 44:23, אָחָפָּאָר 49:3)
```

Identifying the servant as distinct from Israel is practically only possible if it is claimed that Isaiah 40–55 is not the original context of Isaiah 53. This is commonly done by claiming that the four servant songs are originally independent from Isaiah 40–55, as both Janowski (1993) and Hermisson (2017b) do.

Janowski (1993, 8) refers to Steck (1984, 1985). Put briefly, Steck (1984, 372, 381, 387; 1985, 46f) suggests a tripartite content of all songs, treating the mission of the servant, how it is undertaken, and whether it succeeds. Similarly, Hermisson (2017b, 717) argues that the servant songs constitute an independent unit by being bound together by the theme of the servant's success. These arguments are logically not compelling. Since they only show that these four songs have common themes, they do not exclude that there are other passages in Isaiah 40–55 which share the same theme, or which share another theme with (some of) the servant songs. Also the structural arguments of Hermisson (2017b, 717) are not compelling. For example, he argues that after JHWH has called Cyrus from the north in 41:25–29, the original text of Isaiah 40–55 continues with an oracle about Cyrus in 42:5–8*, showing that 42:1–4 must be a later addition. However, it might as well be argued that 42:5–8 thematically belongs to 42:1–4. While it is impossible to analyze the structure and themes of Isaiah 40–55 here, the arguments of Steck and Hermisson do not seem to be compelling.

In addition, it can be observed that in the above list of attributes shared between the servant of the servant songs and the servant Jacob/Israel, the formulations of every attribute differ in at least two ways, for example by switching from first to third person, and switching the suffix from second to first person. Section 2.6 shows that this rhetoric device of varied reference is frequently employed in Isaiah 53. This might substantiate the claim that the servant songs do not constitute an independent textual unit.

However, there are two important objections against the identification of the servant with Israel. First, the servant is depicted as innocent in Isaiah 53, while Israel is depicted as sinful throughout Isaiah 40-55. This objection will be discussed in Chapter 3. The second objection concerns Is 49:5f:

```
4a
         But I said, "I have labored in vain,
         I have spent my strength for nothing (תֹהוּ) and vanity;
 b
         yet surely my cause is with JHWH,
         and my reward with my God."
5
         And now JHWH says,
         who formed me in the womb to be his servant,
         to return (לְשׁוֹבֶב) Jacob to him,
5a\gamma
  δ
         and that Israel might be gathered to him,
         for I am honored in the sight of JHWH,
         and my God has become my strength —
         he says: "It is too light a thing that you should be my servant
6
         to raise up (לָהָקִים) the tribes of Jacob
6a\beta
         and to return (לְהַשִּׁיב) the survivors of Israel;
  Υ
         I will give you as a light to the nations,
         that my salvation may reach to the end of the earth."
```

Apparently, the servant is given a mission concerning Israel. Thus, he must be distinct from Israel.

However, ל in combination with an infinitive "is very often used after a verb to express an action which gives more details about or explains the preceding action" (Joüon and Muraoka 1991, § 1240, their emphasis), for example, the common מֹל after verbs of speech. Thus, verses 5aγδ and 6aβγ might as well be translated as

```
5aβ who formed me in the womb to be his servant,
γ by returning Jacob to him,
such that Israel might be gathered to him,
6a he says: "It is too insignificant for your being my servant by [me] raising up the tribes of Jacob and by [me] returning the survivors of Israel;
```

In this translation, verses $5a\gamma\delta$ and $6a\beta\gamma$ do not give the servant a mission concerning Israel, but describe how JHWH made Israel his servant.

Against this, Hermisson (2003, 354) argues further that verse 4 makes it impossible to interpret the servant as Israel, since it is inconceivable that Israel could complain about having labored in vain for JHWH. However, 49:4 is not necessarily a complaint, nor does it explicitly state that the servant labored for JHWH. The also occurs in 44:9: "All who

make idols are nothing (הוהו)." Thus, 49:4a might rather be a confession: Although JHWH had called Israel to be his servant (49:1–3), Israel had worshipped idols. But now (אָבֶן, 4b), the repenting Israel realizes that its cause is with JHWH.

Thus, the arguments for excising the servant songs from Isaiah 40–55 do not seem compelling. However, even if the hypothesis of the four separate servant songs is wrong, this does not necessarily imply that Isaiah 40–55 is the original context of Isaiah 53. For example, it can be argued that the statement "JHWH has comforted his people, he has redeemed Jerusalem" (52:9b) corresponds to "Comfort, comfort my people, says your God. Speak tenderly to Jerusalem [...]" (40:1f), and that the message of JHWH's return together with his people in Is 52:7–12 corresponds to Is 40:3–11. Thus, it can be argued that Is 52:7–12 concludes Is 40–52:12, such that Is 52:13–Is 55 comprise later additions (Hermisson 2017a, 2.2.).

Obviously, the question of the context of Isaiah 53 is too complicated to be answered here. It seems reasonable to doubt that the servant songs constitute a context separate from Isaiah 40–55, which in turn seems to be the most reasonable assumption for the appropriate context. However, it cannot be excluded that the original context of Isaiah 53 is different than that.

1.3 Conclusion

To the three questions of the context of Isaiah 53, the identity of the servant, and the relation of his suffering to reconciliation, the reviewed interpretations offer only two sets of answers, which ultimately depend on the the supposed context.

If Isaiah 53 is considered integral to at least Isaiah 40–55, the servant is inevitably identified with Israel or a group of Israelites, since the servant is called by the double designation Jacob/Israel both in the majority of occurrences and in the firs occurrence within these chapters. Concerning the relationship to God being broken because of sin, reconciliation is considered to be by the servant's atoning suffering, based on expressions which can be related to cultic expiation of sin.

If, on the contrary, Isaiah 53 is considered one of four independent servant songs, this allows for a different interpretation of the servant of these songs, since the double designation Jacob/Israel does not occur in them. The servant of the songs is commonly considered to be the prophet Deuteroisaiah, who by his death pays for his people's guilt, but ultimately reconciles them by his message, causing contrition or faith.

Chapter 2

The Text of Isaiah 53

It is commonly acknowledged that Isaiah 53 constitutes a unit (Joachimsen 2011, 50).

The text of Isaiah 53 is commonly acknowledged to be well preserved (North 1964, 28). Apart from the Masoretic text (MT) and the LXX, there are two copies, 1QIs^a and 1QIs^b, and several fragments of the book of Isaiah from Qumran (Goldingay and Payne 2006a, 9). 1QIs^a was probably copied between the early second and the mid-first century BC, and presents a tradition slightly different from MT (9, 12). Its text of Isaiah 53 is complete (9). The other Qumran texts were probably copied in the mid-first century AD (9). 1QIs^b "is closely similar to" MT (10). Its text of Isaiah 53 "hat anfangs kleine, ab 53,7 größere Lücken (bis zu ca. 50% der Zeile)" (Hermisson 2017b, 316).

The LXX translation of the book of Isaiah is dated to the mid-second century BC (Goldingay and Payne 2006a, 13). As will be shown in Section 2.5, LXX' text of Isaiah 53 differs distinctively from MT and 1QIs^a, such that the theological content of LXX' Isaiah 53 differs significantly from the Hebrew versions. Since MT and 1QIs^a agree against LXX, it is likely that LXX is emended because of theological considerations.

2.1 Differences between MT and 1QIs^a

The following differences between MT and 1QIs^a have implications for the understanding of Isaiah 53:

14aβ: 1QIs^a has משחתי instead of מְשְׁחָת is ambiguous, since it can be the construct state of both מְשְׁחָת, "anointment," or מְשְׁחָת, "disfigurement." This ambiguity is not resolved by 1QIs^a, since משחתי exhibits the same ambiguity.

In the Qumran scrolls, messianic characters are encountered within the context of the last days. The messiah of Israel is also called *Branch of David*. The commentary on Isaiah 10:34–11:1 interprets this passage as being about the Branch of David, and another scroll describes him in the language of Isaiah 11. (VanderKam and Flint 2002, 265f) Isaiah 11:1 states that the branch shall grow out of Jesse's roots (מָשֶׁרָשֶׁיוֹ), which might be alluded to by בַשֶּׁרֶשׁ in Is 53:2. In accordance with supposing a messianic

interpretation of Isaiah 53 at Qumran, Hermisson (2017b, 318) regards משחתי as derived from משח. Contrary to him, Abegg et al. (1999, 359) regard משחתי as derived from מְּשְׁחָה, "my marring." LXX translates it by ἀδοξησει, "dishonored."

15aβ: 1QIs^a and LXX take עָלִין together with 15aα instead of 15aβ (Hermisson 2017b, 319). That אָלָין most likely belongs to 15aβ, is shown in Section 2.4.1.

3aβ: Instead of the passive participle יודע, 1QIs^a has the active participle יודע (Hermisson 2017b, 322), which is supported by LXX. The passive unambiguously conforms to the parallel אָישׁ מַכְאַבוֹת. The active "knowing sickness" is ambiguous, as this for example could also be said about a physician.

3ba: מְּמֶּנוּ is ambiguous, meaning both "from him" and "from us", resulting in an ambiguous "to the point of hiding faces from him / to the point of him hiding his face from us". Instead of the substantive מֵּסְתֵּר, 1QIsa has מֵסְתִּר, hiphil active participle singular (Hermisson 2017b, 322), thus resolving the ambiguity in favor of the latter. This is also done by LXX.

8bβ: Verse 8bβ was added to 1QIs^a later, probably by a different writer, and has עמי instead of MT's עמי. MT has the lectio difficilior, and is supported by all other textual witnesses (Hermisson 2017b, 326).

Also, apart from Is 53:6, משה as metaphor for Israel is specified only once by משה. This occurrence is in Jer 50:6, where it designates the exiled community. This might indicate that Is 53:6,8 is a reference to Jer 50:6. In Jer 50:6, the exiled community is at the same time called עַמִּי . If Is 53:6,8 is a reference to Jer 50:6, this supports עַמִּי .

9aα: 1QIs^a has plural ויחנו instead of MT's singular for expressing an undefined impersonal subject (Hermisson 2017b, 327), which probably is intended by both versions.

9aβ: Instead of MT's plural בְּמֹחָיו, 1QIs^a has בומתו, "which is unintelligible" (North 1964, 231). It might support singular "his death", which also LXX has. However, the plural most likely is intended, as will be shown in Section 2.6.

10aa: Instead of MT's הָּחֱלִי, hiphil perfect of חלה, 1QIsa has ויחללהו, piel or polel imperfect of חלה, interpreting הַחֲלִי based on verse 5aa (Hermisson 2017b, 329). For the following, it is sufficient that both versions express suffering, and that both versions refer back to words in verses 1–9 (the latter will be used in Section 2.6).

11aαβ: Both 1QIs^{a,b} and LXX add אוֹר as direct object for בְּרָשָׁתוּ. 1QIs^a takes בְּרָשׁתוּ to 11aγ by inserting 1 before it. Concerning both, MT's version seems preferable for the following reasons:

First, as will be shown in Section 2.3, verses 10aβ–11aα allude to Is 41:20, where ראה is used intransitively in its meaning "to understand" (as in English "I see!"). This allusion suggests that האה is used in the same way here.

Second, as will be shown in Section 2.6, the poet in general varies expressions. The variation transitive/intransitive is employed for פֿגע ($6b\beta/12b\beta$). In verse $10a\gamma$, אַד has

as object. This substantiates that יֵרָאָה should not be supplied with an object here.

Third, as will be shown in Section 3.1, Isaiah 53 frequently employs the rhetoric device of ambiguity. Without ו before it, בְּרֵשׁהוֹ can be ambiguously taken together with either 11aa or 11aγ. MT emphasizes this by making בְּרֵשׁהוֹ its own stich, not resolving the ambiguity. Further, the structure of Isaiah 53 (see Section 2.6) suggests that בְּרַשׁהוֹ should primarily be taken together with 11aa, which rules out to separate the two stiches by ז. Taken together with 11aa, יִּיְשָׁבְּע בְרַשׁהוֹ here means "to understand," and thus does not need an object.

Consequently, יְרָאָה should not be supplied with an object, and בְּדַשׁתּוֹ should not be separated from 11aa by inserting ז before it.

12bβ: 1QIs^{a,b}, 4QIs^d, and LXX have "for their transgressions" instead of MT's לַפֹּשְׁעִים, "for the transgressors" (Bons et al. 2011, 2669). Both versions convey that the many are transgressors, which is sufficient for the following. That MT's version most probably is intended, will be shown in Section 2.6.

2.2 Expiation of Sin versus Punishment of Sin

All reviewed interpretations of Isaiah 53 which consider its context to be Isaiah 40–55 interpret the servant's suffering as atoning. This interpretation is based on expressions which can be related to cultic expiation of sin. Therefore, Sections 2.2.1 and 2.2.2 study expressions in Isaiah 53 which can be related to cultic expiation of sin. Since sin alternatively can be condemned and punished, Section 2.2.3 studies expressions related to judgment and punishment of sin.

2.2.1 The Burden of Sin

Sin, the guilt incurred by sin, and the punishment for the guilt constitute a unity which cannot be split up (Knierim 2001, 365). This unity is expressed by

מְטָא denoting both sin and guilt¹

denoting both sin, guilt caused by sin, and punishment (for guilt).² "Metonymic usages of the term illustrate clearly the relationship in Hebrew thought between 'sin' and resultant 'guilt' and 'punishment,' since 'awôn may denote any of these three senses (or all three meanings) in a single passage. In Gen 4:13, for example, it clearly signifies 'guilt' (forensic and psychological) or 'punishment' (penal), and probably connotes both." (Cover 1992, 32) Because of the insoluble relationship between sin and resultant guilt and punishment, the term

gif expresses the continuity and unity

- 1. The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "חָמָא
- 2. The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "עָּוֹן"

of sin and its consequences, a continuity and unity which cannot be partitioned into an act and its various consequences (Janowski 1982, 74).

Zimmerli (1969, 239) suggests that סבל מון and ממא מון, and also סבל מכאב (4aβ) are variations of the "Formel נשא עון." Despite the broad use of in the Old Testament and the variation of this expression in Isaiah 53, Zimmerli (1969, 239f) is convinced that these expressions in Isaiah 53 relate to the priestly use of Zimmerli (1969, 239f) suggests that concerning Isaiah 53, the most important of these priestly uses is the description of the scapegoat and of the sacrificial animal. The scapegoat eliminates the guilt of the people by carrying it away (Lev 16:22). The sacrificial animal atones for the guilt of the people by carrying it (Lev 10:17).

The expressions כבל מכאב, נשא הלי ,נשא הלי ,נשא מכאב are used in the Old Testament as follows:

Apart from Is 53:11b, סבל occurs with עון as object only in Lam 5:7. There, descendants lament that they have to bear the punishment for their ancestors' sins (Lam 5:1-15).

Apart from Is 53:12ba, אים occurs eight times with אים as object. It is always the consequence of a specific misdeed. In six instances, it is followed by a specific penalty justified by the misdeed; the death penalty for blasphemy and idolatry (Lev 22:9, Lev 24:15, Num 18:22, and Ezk 23:49); childlessness for adultery (Lev 20:20), and exclusion from the people for not keeping the Passover (Num 9:13). The two remaining occurrences are in Lev 19:17 and Num 18:32. These verses are commandments of the form "do [...] and you will not bear offence because of it." Consequently, no penalty is specified.

Thus, סבל שון and מבל מון express incurring guilt and being punished for sin. They express bearing the burden of sin.

Apart from Is 53:4aa, נשא הלי occurs only in Jer 10:19, where Jerusalem laments that she has to bear the wound of being deserted by her children. סבל מכאב occurs only in Isaiah 53.

Contrary to Zimmerli's suggestion, נשא המא, סבל עון, and נשא הלי refer to bearing the burden of sin or sickness in all its dire consequences, not to eliminating it nor to atoning for it, as in Lev 16:22 and Lev 10:17, respectively. While נשא עון can express elimination or atonement, נשא חמא and מבל עון unambiguously state that the burden is borne.

2.2.2 Expressions Related to Expiation of Sin

ווָה 15aα <u>יוַ</u>ה

occurs 24 times in the Old Testament. Four times, it occurs in qal and means "to spatter." All other occurrences are in hiphil. Except for Is 52:15, הוו in hiphil occurs as part of expiation of sin, and means "to cause to spatter," that is, "to sprinkle." Since it evokes the context of cultic expiation of sin, 15aa is translated to "he will sprinkle many nations" by proponents of the servant's suffering being atoning, for example, by Sweeney

(2016, 209) and Berges (2015, 212). However, in all hiphil occurrences of מוֹם except Is 52:15, אוֹם has a liquid (blood or water) as direct object, and the target of the sprinkling is specified by the preposition ל, אל, or ל, אל Berges (2015, 214) excuses the lack of a liquid in Is 52:15 by suggesting that in this verse, וווי is used metaphorically. However, this does not excuse the lacking preposition. In general, a verb in hiphil has its direct object as its logical subject. Thus, if ווי here means "sprinkle," it does not mean that something is sprinkled on the many nations, but that the many nations are sprinkled on something. For the sentence to mean "he will cause many nations to sprinkle [something]," the hiphil has to be causative twice, "he will cause many nations to cause [something] to spatter [on something]."

It has been suggested to emend π , or to suppose a secondary meaning "to leap" based on the Arabic nzw meaning "to leap" (Hermisson 2017b, 318). Childs (2001, 412) suggests that the issue might rather be a matter of semantic range. That "to be startled" can be within the semantic range of a verb meaning "to spatter", is attested by the Norwegian "skvette", which has both meanings. Because of the lack of liquid and preposition, the primary meaning of the literal translation to Norwegian, "han vil få mange nasjoner til å skvette," is "he will cause many nations to jump/be startled."

Verse $15a\beta$ describes the kings' astonishment to the point of speechlessness, and verse 15b gives the reasons for their astonishment. The translation "startle" conforms best with this context. Further, verses $14-15a\alpha$ are structured in a chiastic pattern:

In the pattern, יְּשָׁהְמֹּנּ corresponds to שָּׁבְמֹנּ. "He will startle" conforms better with this correspondence than "he will sprinkle."

Finally, "he will startle" is supported by LXX' θαυμασονται, "they will be astonished."

ענה pual 4bβ

in piel means "to humiliate," and accordingly "to become humiliated" in pual. To humiliate oneself (four times in piel, once in pual) is required on the Day of Atonement (Levticus 16 and 23).

$egin{array}{ll} eta a & 8b eta; עוֹן 6b eta, 11b \end{array}$

Both שָּׁיֹשׁ and אָנוֹן are used to designate the sins of Israel that the high priest confesses over the head of the scapegoat (Lev 16:21). The third term for sin in Isaiah 53, אֶחֶא, is not

used in connection to atonement. In Lev 16:21, הַּמָּאה is used. In Lev 5:6 and Lev 5:15, the terms מַעֵּל and אָשֶׁם, respectively, designate sin for which an מַעֵּל has to be presented.

$7aeta\gamma;$ עאָד 6alpha

אָה, a sheep or goat, occurs 47 times in the Old Testament (Waschke 1993, 718). Of these, the following 19 are related to sacrifices:

- Gen 22, two occurrences as burnt offering (שֹלֶה). A burnt offering may (Lev 1:4) or may not (Lev 22:17) be atoning. In Gen 22, atonement is not a concern.
- Ex 12, three occurrences as the Passover lamb, not atoning.
- Ex 13:13, 34:19, 34:20, Lev 27:26 within regulations concerning firstborn animals. These were offered as thanksgiving (Miller 2000, 119), and are thus not atoning.
- Lev 5:7, as אָשָׁם, atoning.
- Lev 12:8, as burnt offering. Since a pigeon is offered as a sin offering together with it, the burnt offering is probably not atoning.
- Lev 22:23, as freewill offering (נְּרָבָה), not atoning.
- Num 15:11, as burnt offering, sacrifice (תְּבָּבֶּ), or freewill offering. In general, the הַּבָּבּ is given as thanksgiving offering, votive offering, or freewill offering (Miller 2000, 113), and is thus not atoning. In this passage, atonement is not a concern.
- Deut 14:4, declared as pure for eating
- Deut 17:1, forbidden as offering if defected
- Deut 18:3, in the definition of the priest's part of the offering
- Is 43:22, as burnt offering.
- Is 66:3, as offering.
- Ezk 45:15, as burnt offering, atoning

Only two of these occurrences are explicitly connected to atonement, while at least 13 occurrences do not seem to relate to atonement. Waschke (1993, 721) suggests that שֵׁל is used less often than בָּבֶשׁ, צֹאֹן, and עֵוּ to designate a sacrificial animal, because the emphasis of the term is on the individual.

However, שָּׁה in general designates an individual of צֹאֹן, a flock (Waschke 1993, 718). commonly designates sacrificial animals (Waschke 1989, 866f). Specifically, צֹאֹן is used as atoning אָשָׁה in Lev 5. In Lev 5:6, the animal has to be female, and in Lev 5:15ff, is is emphasized that the animal should be without blemish.

πבט 7aβγ

קבּב, slaughter, appears twelve times in the Old Testament, but never in the context of cultic expiation of sin. The nine occurrences in Isaiah, Jeremiah, and Ezekiel are all within the context of God's judgment and punishment. In the other three occurrences,

Gen 43:16, Prov 7:22, and 9:2, חבר is used for secular slaughter of animals. חבר occurs only once with שו as object, in Ex 12:37, specifying the punishment for slaughtering a stolen sheep. In the Old Testament, חבר is always related to the profane slaughter of animals for the purpose of a meal (Hamp 1982, 303). While it is likely that in Israel, any kind of slaughter was connected with sacrificial rites (303), this does not imply that any slaughter can be related to atonement. Rather, conforming to rites and rules when preparing or eating meals (for example, not eating blood) was necessary to avoid to sin (1Sam 14:34).

נזז , רָחֵל 7αβγ

Except for in Is 53:7, ewe, appears twice in Gen 31f concerning ewes of Laban and Jacob, and once in Song 6:6. None of these occurrences is related to atonement.

to shear, appears 15 times. In Jer 7:29, Mic 1:6, and Job 1:20, shaving one's head is a sign of lamentation. In Nah 1:12, JHWH "shaves off" the evildoers as part of taking vengeance on Nineveh. Otherwise, 32 is used for regular shearing of sheep as part of sheep keeping. None of these occurrences is related to atonement.

יחול and מו occur together only in Gen 31:19, albeit in a completely different manner: "Now Laban had gone to shear his sheep, and Rachel (בְּחֵל) stole her father's household gods." This only other combination of הָחֵל and מו in Gen 31:19 and the only combination of שׁה with שׁה in Ex 12:37 have theft as common theme. This may indicate that an allusion to a common theme of theft in 7aβ and γ is intended.

8αα לקח

in qal is used for presenting a sacrificial animal, for example, in Lev 12:8, 14:12, and 16:5.

בּשָׂב 10aβ

The word occurs 47 times in the Old Testament. The vast majority of occurrences is within priestly writings, in Leviticus, Numbers, and Ezekiel, where it designates a guilt offering. The priestly writings in Leviticus, Numbers, and Ezekiel, where it designates a guilt offering in 2Kings 12:17 and Ezr 10:19. It designates "compensation [for a committed sin]" in Num 5 (three occurrences) and in 1Sam 6 (four occurrences). Apart from these and Is 53:10, The priestly occurs in Gen 26:10, Jer 51:5, Ps 68:22, and Prov 14:9. In Prov 14:9, none of the meanings of The priestly fits into the context. Most likely, the text is corrupt (Kellermann 1973, 470). In Gen 26:10, Jer 51:5, and Ps 68:22, it means a state of being guilty.

"The term comes originally not from the cultic but from the legal context, from a situation of guilt with the resulting obligation to discharge it" (Spieckermann 2004, 3). The

two key aspects of the Old Testament's use of pipe are "eine Situation der Schuldverpflichtung, in der jemand etwas gibt" and "eine Situation, in der jemand zur Schuldableistung verpflichtet ist oder wird, in der er etwas geben soll" (Knierim 1971, 252f). "Der Primärgesichtspunkt ist die aus einem Urteil folgende Situation der Schuldverpflichtung, des Haftpflichtig-Seins, und ihrer Erfüllung. [...] In diesem Sinne bedeuten dann die Nominalformen [...] das Schuldverpflichtetsein" (254, his emphasis). Since pipe was used for the several aspects of being obliged by guilt, including paying a compensation, it also became a designation for the means of compensation (255). Knierim (1971, 256) suggests that pipe is to be translated as "Schuldpflicht, Haftpflicht, Schuldverpflichtung." Translating it by "guilt offering" is problematic if not wrong, since "offering" implicitly denies the term's aspect of punishment (256).

However, in Leviticus, the אָשָׁה is "most holy" (בְּדֶשׁ קְּדָשׁ לְּדָשׁ לִּדָשׁ, Lev 7:1, 14:13), is ritually slaughtered (שְּלָה, Lev 7:2) at the same place as the burnt offering, its blood is dashed (בְּדָשׁ לִּדְּשׁ, Lev 7:2) against the altar, and its fat is burnt on the altar. The אָלָיִה is declared to be like the sin offering (Lev 7:7). The שְּלָּה is not simply delivered to the priest as a compensation, like the Philistines send their golden שְּלָּה in 1Sam 6. In Leviticus, clearly designates an offering. Thus, while Knierim (1971) has a point that the translation "guilt offering" somewhat denies the term's aspect of punishment, his suggested translations "Schuldpflicht, Haftpflicht, Schuldverpflichtung" certainly deny the sacrificial character of שִּלְּשָׁה in Lev 7. In instances where שִּלְּה designates a sacrifice, a better German translation might be "Schuldpflichtopfer." English seems to lack an equivalent. Maybe liability offering comes closest.

The two verbs used in connection with אָשֶׁבּ in 10aβ are מים and אַכּא. Presenting a sacrifice is never expressed by שׁים, but in general by בוא hiphil (Berges 2015, 268). Presenting an אָשֶׁבּ is expressed by בוא hiphil in all instances in Lev 5, Lev 19:21, Num 6:12, and (in hophal) in 2Kings 12:16, and by שׁוּב in Num 18:9, but by לקח in Lev 14:21, followed by the priest taking (לקח) it.

is used 18 times in the Old Testament, but never in connection with atonement.

Expressions Lacking

Isaiah 53 lacks significant expressions which commonly are associated with cultic expiation of sin and atonement, most notably מבר and the atoning blood ק, and also the verbs for ritual slaughter, זבח and זבח and the atoning blood.

2.2.3 Expressions Related to Judgment and Punishment of Sin

אָשֶׁם (10aβ): "being liable by guilt." See Section 2.2.2.

יהות (1b): In Deuteroisaiah, JHWH's arm often is his instrument of deliverance and of judgment.³

תְּשָׁא (12ba): "offence." "Der Ausdruck 'Vergehen, Verfehlung' steht dem ursprünglich profanen und dem rechtlichen Charakter der Sache grundsätzlich näher als der Ausdruck 'Sünde,' da er die Disqualifizierung einer Tat durch Israel in Form der Rechtskategorie exakt wiedergibt." (Knierim 1965, 67)

מבת (7aβ): "slaughter" as part of God's punishment, see Section 2.2.2

יסר (5ba): "God's punishment." מוּסָר in general means "discipline, chastisement." מוֹסָר aims at fostering proper conduct, which can happen by instruction, rebuke, punishment, or combinations thereof (Sæbø 1971, 739). Similarly, aims at fostering proper conduct, except when used in a juridical context, where it means "punishment" (740). Within the prophetical judgment speeches, מוֹסֶר and מוֹסֶר always mean God's punishing judgment of his people (741). On the one hand, Isaiah 53 is not a prophetical judgment speech. On the other hand, Isaiah 53 constitutes a juridical context, which favors the translation "punishment."

Within Deuteroisaiah, מּוֹסֶר occurs only in Is 53:5. Within the prophets, מוֹסֶר occurs 14 times. God is its source except in Jer 10:8, "the teaching of idols is wood." In Ezk 5:15, JHWH announces to destroy Jerusalem, and thus make her a מוֹסֶר ("lesson") for the nations around. In eight instances, and thus make her a לקח is the object of מוֹסֶר. In these instances, God generally accuses the Israelites for not taking מוֹסֶר, thus מוֹסֶר there means "teaching, instruction." This leaves the three instances Is 26:16, Jer 30:14, and Hos 5:2.

Hosea 5:2 is part of a prophetic judgment speech. מוּסָר here clearly means "punishment." JHWH is speaking through the prophet, thus, מוּסֶר here is God's punishment.

Isaiah 26:16 is not part of a prophetic judgment speech, but of a song of lamentation. Judah's inhabitants lament their distress when JHWH's מּנְּסֶר was on them. The distress is likened to that of giving birth. Thus, מּנְסֶר here can hardly have positive connotations, and means "punishment." The verse states JHWH as the source of the punishment: "JHWH, [...] your punishment was on them." Thus, מּנְסֶר here is God's punishment.

Jeremiah 30:14b is the instance most similar to Is 53:5:

I struck you (הָּכִּיחִיךְ, from הָבָּיחִיךְ, in Is 53:4bβ) with an enemy's strike with the punishment (מוֹפַר) of someone cruel because your misdeed (עֵוֹנֶךְ) is great, your sins (חַמֹאֹחָיִר) are numerous.

JHWH is speaking, thus, it is implied that מוּסָר is God's punishment.

Since God is the source of מוֹסָר in all its occurrences within the prophets except for Jer 10:8, where it is explicitly stated that the source are the idols, it is likely that God is the source of מוֹסָר also in Is 53:5. In Ezk 5:15, JHWH's מוֹסָר is not directed at Israel,

^{3.} The Brown-Driver-Briggs Hebrew and English Lexicon, s.v. "II זרע 1.c"

^{4.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "אָמָא 1."

but at the neighbouring nations. In all other instances where it is not the object of לְקְח means "God's punishment." This suggests that "God's punishment" is one of the meanings of מוֹסֶר in Is 53:5. Like many other expressions in Isaiah 53, מוֹסֶר most likely is ambiguous, and also means "chastisement." This is discussed in Section 2.3.

מְשָׁפְּם (8aa): "judgment."

پاۋ (6b β , 11b): "sin." The term comprises both sin, guilt incurred by sin, and punishment for that guilt.

לאָר (8aa): "arrest." The verb עצר is used for putting in prison or being kept in the court of the guard in 2 Kings 17:4, Jer 33:1, and Jer 39:15. In all instances, the one imprisoning is a law enforcement authority (the king or his guards), implying that שֹבֶּר means arrest.

שנה pual (4b β): "to become humiliated." The cultic use of שנה is closely linked to juridical language (Gerstenberger 1989, 254). "Rechtliche und kultische Verwendung von ' $\bar{a}n\bar{a}h$ II pi/pu greifen so stark ineinander, daß die Priorität eines Bereiches noch fraglich ist. [...] Hauptvertreter des Verbs ist der pi-Stamm. Er hat rechtlich-kultische oder kultisch-rechtliche Bedeutung" (255f).

ששע (5aα, 8bβ), פֿשָּעם (12aδ, 12bβ)): "felony," "acts which break relationships within the community and with God." The term is strongly forensic (Knierim 1965, 179). Since JHWH is the law-giving authority, every felony constitutes an act against God (182) and provokes JHWH's judgment (184). In Isaiah 40–55, the emphasis is on the severity of the act and its consequence of breaking the relationship between JHWH and Israel or between persons (183).

צרק (11aγ): צרק in hiphil occurs twelve times in the Old Testament. In the other eleven occurrences, it has a forensic meaning, "to obtain rights for", "to admit as right", "to declare as innocent", "to treat as innocent", and "to assist someone towards his rights."

צַּדִּיק (11aγ): "juridical; persons whose conduct will be checked and found irreproachable, innocent, in the right." The root צדק has a twofold meaning. On the one hand, it designates forensic righteousness, that is, conformity with the given norm. God judges this conformity and rewards and punishes accordingly. On the other hand, it designates a right relation to God, almost a synonym for salvation. (Johnson 1989, 903)

שֶּׁה (7aβ): "lamb." Six occurrences as being judged by God (Ezk 34:17,20,22).

2.2.4 Conclusion

The expressions נסא חמא seem to belong to the framework of judgment and punishment of sin rather than to that of cultic expiation of sin. Thus, the expressions

^{5.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "פַּשָּׁעַ 2."

^{6.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "צרק hif."

^{7.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "צַּדְּיִקְ 2."

related to cultic expiation of sin in Isaiah 53 are ענה, עוֹן, נוֹה, לקח, אָשָׁם pual, צֹאן, and מנה, עוֹן, נוֹה, לקח, אָשָׁם, a total of ten expressions.

The expressions related to judgment and punishment of sin in Isaiah 53 are סבל עון and איש, and the expressions listed in Section 2.2.3, a total of 17 expressions. Even though אָשֶׁם occurs mostly in relation to cultic expiation of sin, thus supporting a cultic framework for Isaiah 53 more than a juridical framework, the expressions in Isaiah 53 seem to relate primarily to judgment and punishment rather than to atonement.

2.3 Expressions Related to Knowing God and Keeping His Commandments

Committed sin is dealt with by expiation or punishment. However, sin can also be dealt with preemptively, by avoiding to sin. This is achieved by appropriate fear of God, and corresponding righteous conduct. Isaiah 53 contains a comparatively small, but significant number of expressions related to knowledge of God and keeping his commandments. These are:

שׁכל (13a): "to have insight." The majority of occurrences of the root is within wisdom literature (Koenen 1993, 785). שׁכל conveys both secular prudence and recognition of, faith in, and appropriate fear of JHWH. This acknowledgement of JHWH leads to keeping his commandments (786f).

ראה (15ba, 11aa): "to understand." In its theological use, ראה comprises the entire spectrum of visional encounter between humans and God (Fuhs 1993, 250). Most occurrences refer to experiencing God's presence through his actions in history (252f). This experience leads to analyzing one's own actions and conduct, in order to turn the personal relationship, which is based on the act of ראה, into reality (254).

בין hithpael (15bβ): "to consider diligently, understand."

מוּסָר (5ba): "chastisement." On the one hand, מוּסָר means a content to be learned. In the theological realm, it comprises knowledge necessary to lead a life pleasing to JHWH (Branson 1982, 692f). On the other hand, מוּסָר means a method of instruction (694). If its source is JHWH, it is in most cases redemptive by restoring the chastised to proper conduct (694f).

דְּעָת (11aβ): "knowledge." דְּעָת occurs 40 times in proverbs (Bergman and Botterweck 1982, 496). These occurrences can be partitioned into older occurrences, where דְּעַת refers to secular knowledge, and younger occurrences, where דְּעַת is closely related to fear of JHWH (496). For these younger occurrences, דְּעַת means acquaintance with God and walking in his ways (497). It comprises a veneration for God, which constitutes itself in righteousness and piety (497). דְּעַת יהוה is also paralleled with יַרְאָּת יהוה is 11:2.

דעת can also mean "knowledge of God." "To know JHWH" indicates a relationship with JHWH, who wants to deliver and protect those who know him (Bergman and Botterweck 1982, 499).

צָּדִיק (11aγ): a pious person, who delights in JHWH's commandments, thus behaves according to them, and thus belongs to the community of JHWH.⁹

שים (10aβ) can also have an intellectual meaning of "to direct attention to" (Vanoni 1993, 770). As will be shown in Section 2.4.1, this is probably not meant here. However, the expressions in this section together with יד and יד (both in verse 10b) allude to Is 41:20:

```
so that all may see (יְרָאָד) and know (יְיֵדְעוּ),
all may direct attention to (יְנִשִּימוּ) and understand (יְרִיהוּה),
that the hand of JHWH (יִר־יהוה) has done this,
the Holy One of Israel has created it.
```

The verse concludes a passage assuring Israel of JHWH's help. In this verse as in the Old Testament in general, knowledge of JHWH's deeds mainly concerns his deeds of salvation (Vanoni 1993, 788). The allusion to Is 41:20, where אור ווי is used intransitively, substantiates that אור באה in 53:11aa should not be supplied with an object.

2.4 Translation

Isaiah 53 exhibits several ambiguities. The translation lists their various meanings.

```
13a
         Behold, he will succeed,
                                      my servant,
                         understand,
  b
            he will be exalted, lifted up, and be very high.
14a\alpha
         Such as they were appalled at you, the many —
   β
               such a disfigurement from that of man were his looks
                    an anointment beyond
  b
            and his appearance from
                                        that of a human —
                                beyond
15a\alpha
         so he will startle many nations
   β
               because of him, kings will shut their mouths
  b\alpha
            because what had not been recounted to them, they saw
   β
               and what they had not heard, they understood.
 1a
         Who believed what we heard?
            and JHWH's arm, to whom was it revealed?
  b
```

^{8.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "דַּעָח I 3.a"

^{9.} The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "צַּדִּיק 5."

$2a\alpha$	He grew like a sapling before him
	on his own
β	and like a root from dry earth
Υ	he had neither appearance nor majesty
b	that we should look at him, nor had he good looks that we should take pleasure in him.
$3a\alpha$	Despised and lacking men
β	a man of pains, known by sickness
$b\alpha$	to the point of faces' hiding from him
	his face's hiding from us
β	despised, such that we did not value him.
$4a\alpha$	However, our sicknesses he bore
β	and our pains, he carried them
$b\alpha$	whereas we evaluated him
β	struck, smitten by God, and humiliated.
$5a\alpha$	But he was pierced because of our felonies
β	crushed because of our sins
$b\alpha$	God's punishment for our peace was on him
	which we deserved as retaliation
	God's chastisement for our peace was on him
β	by his slash we were healed.
	his being allied
$6a\alpha$	We all like a flock of sheep went astray
β	each to his own way we turned
$b\alpha$	but JHWH pushed off on him
β	the burden of the sin of us all.
$7a\alpha$	He was oppressed but he submitted and did not open his mouth
β	like a sheep he was led to slaughter
Υ	and like an ewe before her shearers is silent
b	he did not open his mouth.
$8a\alpha$	From arrest and because of judgment he was taken
β	but as for his contemporaries, who pondered
$b\alpha$	that he was cut off from the land of the living
β	because of the felony of my people, he was struck?
$9a\alpha$	One assigned with the wicked his grave
β	but he was with a rich in his violent death
$b\alpha$	because no violence he had done
	although
β	and no deceit in his mouth.
$10a\alpha$	And JHWH willed crushing him, made sick

β if you appoint as liable his soul liability offering he will see offspring, he will prolong days Υ $b\alpha$ and the will of JHWH, by his hand will succeed. Because of the trouble of his soul he will see, he will be satisfied $11a\alpha$ β by his insight sweat a righteous one will assist my servant towards his rights concerning the many Υ reconcile my servant with God before the many one will assist my righteous servant towards his rights concerning the many the righteous one, my servant, will assist the many towards their rights reconcile the many with God prove himself to be righteous to the many b and their sins' burden he will carry. $12a\alpha$ Therefore, I will give him a share in the many β and with strong ones he will share spoil in recompense for that he stripped to death his soul γ δ and was counted among felons $b\alpha$ but he bore the burden of the many's offence β and will push on behalf of the felons.

2.4.1 Comments

יַשָּׂכִיל 13a יַשָּׂכִיל

The parallel 13b makes "he will succeed" the primary interpretation.

14aα עֶלֵיךְ

י אָלִיךְ is commonly and recently emended to עָלִין (Hermisson 2017b, 317, 314). Against emendation, it is commonly argued that MT has the lectio difficilior, and that prophets commonly switch between second and third person. Concerning the former, Hermisson (2017b, 317) argues that since עָּלִיךְ is an impossible reading, this is not a case of lectio difficilior, but a scribal error. Concerning the latter, he argues that prophets do not switch that abruptly (317). Nevertheless, עָלִיךְ should most likely not be emended, albeit for a different reason; see 15aβ below.

מַשְׁחַת 14aβ

מְשְׁחָת can be the construct state of both מְשְׁחָת, "anointment," or מְשְׁחָת, "disfigurement." מְשְׁחָת occurs 21 times in the Old Testament, but always together with מְשָׁקוּ. For מְשְׁחָת, this would be the only occurrence in the Old Testament. The verbal roots do not resolve the issue. משׁח and משׁח are used 142 and 70 times, respectively, and שִׁמֶּן can be used without שֶׁמֶּן, for example in Is 61:1. That the ambiguity is intentional, is substantiated by the chiastic structure of verses 14-15aa:

In the pattern, מְשְׁחֵת is the only expression which does not have a corresponding expression. This suggest that מִשְׁחַת is regarded as its own corresponding expression by virtue of its double meaning. This in turn is substantiated by 14aα describing the many's negative horror over the servant, whereas 15aα describes their positive astonishment, as made clear by the parallel 15aβ. While their horror is caused by the servant's disfigurement, his anointment might be a reason for their astonishment.

15aα הַיַּ

See Section 2.2.2.

15aβ עָלָיו

Against MT, שָּלִין is sometimes taken together with 15aa, for example by 1QIsa, LXX, and recently, by Joachimsen (2011, 92). However, judging from the chiastic construction of verses 14-15aa, verse 15aa ends with רַבִּים, corresponding to 14aa.

Further, splitting verse 15a between לְבִּים and עָּלָין is in line with several other rhetoric devices applied to these two words in order to emphasize the content of verse 15. First, the order of the two words is reversed relative to עָּלֶיךְ רַבִּים in 14aα. This emphasizes the reversal from negative horror in verse 14 to positive astonishment in verse 15a. Second, several unexpected formulations underline the unexpectedness of what is described: Verse 15 has עָלֶיךְ instead of the expected עָלֶיךְ (supporting that עָלֶין in verse 14 should not be emended to עָלֶין (אֶלֶין has different meanings in verses 14 and 15. Finally, as they correspond to עָלֶין in 14aα, the two words עַלִין are expected to be part of the same stich, but they are not; עָלֶין belongs to 15aβ.

2aα לְפַנֵיו

The last previously mentioned third person singular is JHWH, thus one meaning is "before him [JHWH]." In addition, "das Suff. 3. Pers. Sg. in M, ebenso $Q^a Q^b A S$, könnte sich auf den Knecht selbst beziehen ('für sich hin' [...]); die zu erwartende Disqualifikation wäre dann die Isolation des Knechts" (Hermisson 2017b, 320). This interpretation conforms with "lacking men" in $3a\alpha$.

3bα כַמַסְתֵּר פַנִים מְמֵנוּ

Since verses 3f describe the we-group's negative assessment of the servant, "[us] hiding our faces from him" most likely is the primary interpretation.

מוסר 5bα

As shown in Section 2.2.2, מוּסָר most likely means "God's punishment." The meaning "punishment" is substantiated by the parallel תַּבּוּרָה, "slash." Thus, "God's punishment" most likely is the primary meaning.

In general, מּוֹסֶר means "discipline, chastisement," leading to insight which itself can be designated by the transparation that the servant will understand (13a, 11a). As shown in Section 2.2.2, within the prophets, God is the source of מוֹסֶר except for Jer 10:8, where the idols are stated as its source. Thus, God is most likely the source of chastisement also here.

שַׁלוֹם בּ5bα

in general means wholeness, abundance, peace. Here, in contrast to מוֹסָר, "God's punishment," it probably means "peace with God". Within the construction מּוֹסַר שְׁלוֹמֵנוּ, the word מְּלוֹמֵנוּ can also be interpreted negatively as "retaliation", yielding "the punishment we deserved as retaliation" (Gerleman 1976, 930). Given the parallel "healing for us" (5bβ), the primary meaning is most likely "peace with God."

אַבָּרָתוֹ βבּ

הַבְּרָתוֹ is commonly derived from הַבּּרָת, "slash," but it also is the passive participle from הבר. The female passive participle expresses the abstract "being allied." The context suggests "slash" as primary interpretation.

הַפָּגִּיעַ בּוֹ אָת עַוֹן 6b

"'stoßen' kommt der Grundbedeutung und den Anwendungsmöglichkeiten von hebr. $p\bar{a}\bar{g}a'$ noch am nächsten" (Maiberger 1989, 502). The twofold meaning of in Is 53:6b α

and $12b\beta$ can be imitated using "to push." Concerning $12b\beta$, "to push" can mean "to exert oneself continuously, vigorously, or obtrusively to gain an end." Concerning $6b\alpha$, the American idiom "to push something off on somebody" means "to place one's task onto another person; to make someone else do an unwanted job."

The two expressions נשא חמא attest that the consequences of sin are compared to a physical burden which must be carried, and thus also can be pushed. "He pushes the burden off on him" captures the causative sense of the hiphil of פגע ב, since one makes something push into someone by pushing it onto him.

8aa מַעֹצֵר וּמְמִשְׁפַּט

North (1956, 124) argues that since אוֹף has three meanings (away from, because of, without), there are three possible translations of this expression, implying that אוֹף must have the same meaning in both composites. Likewise, all translations collected by Joachimsen (2011, 385) translate אוֹף to have the same meaning in both composites. Obviously, it is strongly expected that אוֹף has the same meaning in both composites. However, just as the unexpectedness of the content is emphasized by using אוֹף in two different meanings in verses 14f, אוֹף here might be used in two different meanings to emphasize the unexpectedness of the content. If a righteous is arrested and subsequently tried, the righteous is expected to be acquitted. Contrary to what is expected, the servant is sentenced to be taken from the land of the living.

9aβ בִּמֹתָיו

The word is commonly emended, since the literal "in his deaths" is considered meaningless, for example by North (1956, 125). It is emended both to singular or to different words, usually by attempting to find a parallel to "grave" (125f). That מוֹת most likely is intended, will be shown in Section 2.6. מוֹת in plural is also found in Ezk 28:10, where it is a plural of intensification (Hermisson 2017b, 328). Also here, it is a plural of intensification, "his violent death."

שַל שd9

can mean both "although" and "because." "Although" relates to 9aα, "because" to 9aβ.

^{10.} Merriam-Webster OnLine, s.v. "push," accessed October 4, 2017, https://www.merriam-webster.com/dictionary/push

^{11.} McGraw-Hill's dictionary of American idioms and phrasal verbs, s.v. "push something of on(to) someone"

10aα, 10b חפץ

In the context of Is 40-54, אפץ refers "not to a reactive delight but to a proactive determining" (Goldingay and Payne 2006b, 318).

10αβ תַּשִּׁים אָשָׁם נַפְשׁוֹ

This expression has been notoriously difficult to interpret, and as a result, various emendations have been suggested (Joachimsen 2011, 386–390). It is reasonable to first try to interpret MT, which is supported by 1QIs^a.

As shown in Section 2.2, the expressions in Isaiah 53 seem to relate primarily to judgment and punishment of sin rather than to atonement. Thus, "liable" most likely is the primary meaning.

The second person male singular "you" may be understood to be addressing God, but since God is referenced in third person in 10aa and 10b, this seems unlikely. Thus, the "you" seems to address the one hearing or reading the poem.

10aβ-b [...]אָם־

In this context, $\Box \aleph$ can introduce two types of clauses: A conditional sentence, or an optative clause. An optative clause ("If you would appoint his soul as liable!") would make verse $10a\gamma b$ an unconditioned statement. As such, verse $10a\gamma b$ would logically belong to the unconditioned statements following from verse 11. MT's separation of verses 10 and 11 conforms with the overall structure of the poem (see Section 2.6), according to which verse $10a\gamma b$ belongs to verse $10a\beta$. Thus, verse $10a\gamma b$ is most likely the apodosis of a conditional sentence introduced by $\Box \aleph$ in $10a\beta$.

דַעַת 11aβ

means "insight," but can also mean "sweat." The meaning "insight" is substantiated by the parallel יְרָאֶה, he will understand. Also, for the meaning "sweat," this is the only occurrence in the Old Testament. Thus, "insight" most likely is the primary meaning.

11aγ יַצְהִיק עַבְהִי לָרַבִּים

Both deletion (following three manuscripts) and transposition of צָּדִיק have been proposed. While North (1956, 126) supported deletion of צָּדִיק as a dittograph of צָּדִיק, North (1964, 232) argues that such a mistake hardly could "have been perpetuated in the versions and in all but three Hebrew MSS." It is most likely that the mistake is with those three manuscripts, having omitted צַּדִיק by homoioteleuton (232). Transposition "is theologically motivated and lacks manuscript support" (Sapp 1998, 174).

The verse is ambiguous both because of the various meanings of צדק in hiphil, and because it is unclear who is the subject of the sentence.

צדק in hiphil occurs twelve times in the Old Testament. In the other eleven occurrences, it has a forensic sense of "to obtain rights for", "to admit as right", "to declare as innocent", "to treat as innocent", and "to assist someone towards his rights". The hiphil can also be intransitive, "[...] will prove himself to be righteous." Since צַּדִּיק also designates a "right" relation to God, צַדִּיק can also have a causative meaning concerning this relational meaning of צָּדִיק, "will cause the 'right' relationship with God."

The most straightforward interpretation of 11aγ is צַּדִּיק as subject and עַּבְּדִּי as object: "A righteous will cause my servant to be righteous before the many."

However, the straightforward interpretation of 11aγ leaves unanswered who could be the subject, "a righteous." It cannot be the servant Israel, who is the one who is made righteous in 11aγ. It cannot be anyone of the we-group, who confess that אָלָּבֶּי have sinned (6a, 6b). It can hardly be any of the many nations. It could be God, but since God refers to himself in the first person both by the first person suffix of עַּבְּדִּי and by the first person in 12aα, it seems unlikely that he refers to himself in third person here. Thus, there seems to be no righteous subject available.

This leads to considering secondary interpretations of the sentence. One alternative is to take צַּדִּיק to belong to the object שָּבְדִּי , resulting in an impersonal subject, "one will cause the righteous servant to be righteous before the many."

Another alternative is to take the subject to consist of both עַבְּדִי and עַבְּדִי, such that נְבְּדִי is apposition to צַּדִיק. Sapp (1998, 173) defends this interpretation against the objection that צַּדִיק lacks the definite article, stating that this is common, and that the apposition functions as a definite article.

Thus, there are three possible subjects, צַּדִּיק, impersonal, or צַּדִּיק עַבְדִּי can be combined with the forensic or relational meaning of צַדִּיק עַבְדִּי .צרק can additionally be

combined with the intransitive meaning, since there is no direct object left in the sentence. An impersonal subject can only be combined with the forensic meaning, since a righteous servant is not in need of reconciliation. This results in six possible interpretations:

- "A righteous one will assist my servant towards his rights concerning the many." (or any other forensic meaning, for example "[...] will declare my servant as righteous before the many.")
- "A righteous one will reconcile my servant with God before the many."
- "One will assist my righteous servant towards his rights concerning the many." (or another forensic interpretation)
- "The righteous one, my servant, will assist the many towards their rights."
- "The righteous one, my servant, will reconcile the many with God."
- "The righteous one, my servant, will prove himself to be righteous to the many."

Some of these interpretations seem to explicitly contradict each other, with some amounting to the servant being in need of reconciliation, and others amounting to the servant being righteous, and thus not in need of reconciliation. This is discussed in Chapter 3.

יַפְגִּיעַ 12bβ

See $6b\alpha$ above.

2.5 The Text of the LXX

The text of LXX' Isaiah 53 differs distinctively from the texts of MT and 1QIs^a, such that the theological content of LXX' Isaiah 53 differs significantly from the Hebrew versions. The emendations of LXX have at least two effects:

- God is not depicted as willing and causing the servant's suffering, but as wanting to save and vindicate him (4b, 5bα, 6b, 9a, 10aα, 10b-11a)
- The we-group is depicted as siding with God and the servant throughout the events reported (1a, 2b-3, 4b, 8aα)

1a LXX adds xupix at the beginning of verse 1. Thus, it depicts the speakers as siding with God, and thereby, with the servant.

2b–3 LXX translates וְנְרְאֵּהוּ as indicative instead of as indirect cohortative. LXX interprets אַרְאָהוּ as οὐδε καλλος, resulting in translating verse 2b as "and we saw him, and he had no good looks nor beauty." Further, LXX adds ἀλλα το εἰδος αὐτου at the beginning of 3aα, and translates שׁנְיָּהָ by ἀτιμον, and וַחֲדֵל אִישִׁים by ἐκλειπον παρα παντας ἀνθρωπους. Both ἀτιμον and ἐκλειπον are neuter, and therefore attributes to εἰδος. This

results in LXX translating verse 3aα as "but his looks were unhonored, failing behind those of humans." LXX resolves the ambiguity of מָּסְתֵּר פָּנִים מִמְּנוֹ in 3bα in favor of "him hiding his face," maybe because its Hebrew original has מסתיר like 1QIsa. Finally, LXX changes the first person plural הַשַּׁבְּנָה to third person singular passive ἐλογισθη: "he was not valued." Thus, LXX removes all references to the we-group despising the servant.

4b: LXX omits אֱלֹהִים, resulting in "And we considered him to be in distress and under a stroke (of misfortune) [ἐν πληγη] and under oppression [ἐν κακωσει]. The final word in the LXX suggests violent opposition by men, as opposed to humbling affliction allowed by God according to the final word of the Hebrew. [...] LXX has removed any suggestion of divine intent in the Servant's misfortune" (Sapp 1998, 176, his emphasis). The speakers are depicted as sympathetic to the suffering and oppressed servant.

5bα As shown in Sections 2.2.2 and 2.4.1, מוֹסָר here primarily means "God's punishment." LXX translates מוֹסָר by παιδεια, which corresponds to the more positive meaning "chastisement."

6b, 12bβ LXX conforms these two verses to each other. It has plural ταις άμαρτιαις instead of MT's singular מוע in 6bβ, and τας άμαρτιας instead of MT's in 12bβ. פונע is in both verses translated by παραδιδωμι. LXX has passive in 12bβ, maybe due to its Hebrew original having יפּגע like 1QIs^a (Hermisson 2017b, 333): "And the Lord delivered him for/because of our sins" (6b), "and he was delivered because of our sins" (12bβ).

8aa: LXX has פֿע דף דע דמתבוישסוג instead of מֵלֹבֶּר, and continues הָ אַטְוּסג מטֿדסט אָף אָן, and eleting מָלַבָּר, and adding "his" to it, resulting in "When he was humiliated, his right was taken away". Thus, LXX substitutes the reference to arrest and condemnation by humiliation.

9a: LXX translates "And I will give the wicked instead of his grave and the rich instead of his death", changing the verb's third person to first person (implying God as subject), changing the וואס הייים, and changing בו to מֹעִינו before מֹעִינו. The result depicts God as vindicating the servant and saving him from death.

10aα: LXX interprets דכה here, but not in verse 5aβ, based on the Aramaic meaning "to cleanse", and interprets החלי as substantive, resulting in "The Lord desired to cleanse him from sicknesses". "The effect of the LXX's translation is to avoid identifying the Servant's suffering with the Lord's will. The Lord wants to save the Servant from his unjust suffering" (Sapp 1998, 180, his emphasis).

10b-11a: LXX makes all verbs dependent on וְחֵבֶּיּ יהוה, "And the Lord desired ..." Instead of בְּיָרוֹ יִצְּלָח, LXX has ἀφελειν, "to take away," which it takes together with 11aα: "to take away from his soul's agony." LXX interprets יְלְּאָה as hiphil, and adds φως like 1QIsab, "to show him light." LXX translates by πλασαι. As a result, LXX translates verses 10b-11 as "And the Lord desired to take away from his soul's agony, to show him light, and to form by understanding, to make righteous the righteous one, who serves many well." In this version, God wants to relieve the servant from his suffering and to vindicate him.

2.6 Structure

While there is disagreement about the structure of Isaiah 53, it is commonly agreed that Isaiah 53 starts and ends with speeches by God (Hägglund 2008, 46). Because of עַּבְּדִּי in verses 52:13a and 11aγ, God speaks from 52:13, and in 11aγ. God's first speech ends with 52:15, with the we-group speaking from 53:1. The first person singular אַחַלֶּכְ in 12aα suggests God as speaker. Further, רְבִּים in verses 11aγ, 12aα, and 12bα refer back to רַבִּים in verses 14aα and 15aα, supporting the impression that God speaks in 11aγ–12. Whether God's second speech starts before 11aγ, is debated (Koole 1998, 262). Since JHWH is referred to in the third person in 10aα and 10b, God's second speech cannot start before 11aα. Thematically, verses 11aγ–12 correspond to verses 13–15 by relating the future glory of the servant to his suffering.

Further, Isaiah 53 is partitioned into four parts corresponding to the third party figuring besides the servant and God:

13–15 the many 1–9 we 10 you 11–12 the many

A striking structural element is given by the two pairs of repeated consecutive words in 10–11aα, יַבְּשׁׁ יִרְאֶה (10aα) in combination with יְהַפֶּץ יהוֹה (10b), and נַּפְשׁׁוֹ יִרְאֶה 10aβγ and 11aα. The first, יַבְּשׁׁ (10aα), and the last, נַפְשׁׁוֹ יִרְאֶה (11aα) frame the part of the text where "you" is the third party. This suggests that these word pairs function as partition markers or bridges.

That verse 10aα constitutes a bridge between verses 1–9 and 10aβ–b is substantiated by it being composed of references. On the one hand, יְהַפֶּץ יהוה יְהַפֶּץ refers forward to יְהַבֶּץ יהוה (10b). On the other hand, יְּהָאוֹ הָחֲלִינוּ (5aβ), דֹּלָי (5aβ), חֹלִי (3aβ), and יְּלְיֵנוּ (4aα). Likewise, verse 11aα seems to constitute a bridge between verses 10aβ–b and 11f, with referring back to יְּרָאָה in 10aβγ. This is substantiated by that both bridges

correspond to each other not only by referring to two consecutive words in 10aβ–10b, but also, that for both bridges the reference is with two variations: The order of words is reversed from 10aα to 10b, and וופא is used once as a noun and once as a verb. While is repeated verbatim in 11aα, the two words are allotted to different stiches, and יִרְאָּה is used in two different meanings, with and without direct object (supporting that no object should be supplied in 11aα). Thus, the structure can be refined as follows:

13-15 the many 1-9 we $10a\alpha$ bridge $10a\beta-10b$ you $11a\alpha$ bridge $11a\gamma-12$ the many

It is commonly acknowledged that the emphatic אָבֶן in the beginning of verse 4 marks a division of verses 1–9, with verses 1–3 and 4–9 reporting the we-group's previous and current assessment of the servant's suffering, respectively. The reversal of their opinion of him is emphasized by the reversed order of מַלְאֹב in 4a relative to their order in 3aβ, and by the grammatical reversal of the two words: מַלְאֹב has a feminine ending in verse 3aβ, but a masculine ending in verse 4aβ, and מֹלְי is singular in verse 3aβ, but plural in verse 4aα.

Considering these two parts, verses 1–3 and 4–9, it can be observed that verses 1–3 refer to the two word pairs יְהַנֶּץ יהוֹה/יהוֹה חָפֵּץ of the two bridges 10aα and 11aα. Also these references are with variation: יְּהוֹה (one word repeated), and בְּלָּאָה in verse 2b refer to another word of the same root). The two references correspond to each other in that each reference refers from a physical aspect (מַרְאָה, וְּרוֹעֵ) to a psychic aspect (נַפְּשׁוֹ חָפֵּץ) of God and the servant, respectively.

Thus, it is likely that verses 1–3, like verses 10aα and 11aα, function as a bridge, in this case, between verses 13–15 and 4–9. This is supported by לְּשֶׁבְּעָהוּ in verse 1a and in verse 2b referring back to בְּאָבֶין in verse 15b, בְּאָרָן in 2aβ referring forward to מֵאֶרֶן in 8bα, מֹלְיָבֶוּ in 8bα, מֹלְיָבֶוּ in verse 2aγb referring back to מֵאֶרֶן in verse 14aβb, and מַרְאֶהוּ וְחַאֲּרוּ and מֵרְאָהוּ in verse 2aγb referring back to מֵלְיבִינּוּ חָשֵּׁבְנָהוּ in verse 14aβb, and חַלְּבִינוּ חָשַׁבְנָהוּ no verse 3 referring to מֵלְיבִינּוּ חָלְיבִינּוּ חָלְיבֵינּ הוֹ חַשַּׁבְנָהוּ hand הַשַּׁבְנָהוּ the by repetition and the unusual regular plural doubly emphasized אִישִׁים אִישׁים אִישׁ in verse 3a refers both back to מֵּאִישׁ in 14aβ and forward to אַישׁים in 6aβ. Also all these references are with variations:

שמע	1a	+ noun	$15\mathrm{b}\beta$	לא $+ \operatorname{verb}$
אֶָרֶץ	$2a\beta$	$\hbox{``ground''} + \hbox{adjective}$	8bα	"land" $+$ substantive
ראה	2b	imperfect, cohortative, suffix	$15 \mathrm{b}\alpha$	perfect, indicative, no suffix
מַרְאֶה, תֹאַר	$2a\gamma b$	no suffix, separated	$14a\beta b$	suffix, consecutive
אָישִׁים אִישׁ	3a	repetition, plural	$14a\beta$	preposition
		"all men, a man"	$6a\beta$	"each man"
מַכְאֹב	$3a\beta$	female, no suffix	$4a\beta$	male, suffix
חֹלִי	$3a\beta$	singular, no suffix	$4a\alpha$	plural, suffix
חֲשַׁבְנֻהוּ	$3b\beta$	negated	$4b\alpha$	with predicative accusative

The first three pairs of references (מַרְאָה, תֹאַר), (ראה, תַּמְרָאָב), and (חֹלִי, מַכְאָב) correspond to each other in that the order of words is reversed in the second occurrence of the pair.

The three bridges separate Isaiah 53 into four main units, verses 13-15, 4-9, $10a\beta-10b$, and $11a\gamma-12$. These units correspond thematically to each other by describing the consequence of the servant's suffering for the third party (and the servant).

That this might be the structure of Isaiah 53, is further substantiated by verses 11aγ–12 referring not only back to 52:13–15, but also to verses 4–9. Except for which is hard to vary, also these references are with variations:

پاز	11b	plural, suffix	$6b\beta$	singular, no suffix
הוא	$11b,12b\alpha$		$4a\alpha$, $5a\alpha$, $7a\alpha$	
סבל	11b	imperfect, no suffix	$4a\beta$	perfect, suffix
מות	$12a\gamma$	+ singular, no suffix	$9a\beta$	\Rightarrow + plural (supporting MT), suffix
פשע	$12a\delta,12b\beta$	פֿשָׁעִים $+$ ל $/$ אָת	$5a\alpha$, $8b\beta$	פָשַׁע + מָן
פגע	$12b\beta$	imperfect, intransitive	$6b\alpha$	perfect, transitive

ירוּעַ in verse 11aβ refers back to יִרוּעַ in verse 3. This suggests that from the structural viewpoint, verse 11aβ belongs to the bridge 11aα. This substantiates that בְּרַשְׁתוּ primarily belongs to יִשְּׁבְּע and should not be separated from it by inserting i before יִשְּׂבְע as 1QIsa does. The resulting meaning "he will be satisfied by his knowledge" substantiates as a parallel to יִרְאֶה in 11aα, that יִרְאֶה means "he will understand" and thus does not need to be supplied with an object.

Thematically, verses $10a\beta$ –10b correspond to verses 13–15, 4–9, and $11a\gamma$ –12 by describing the consequence of the servant's suffering for the third party (and the servant). However, verses $10a\beta$ –10b are rhetorically set apart from the remainder of Isaiah 53 by being a conditioned statement.

Thus, the overall structure is as follows:

A	13 - 15	The consequence of the suffering for the servant
B^1	1–3	bridge
С	4-9	The consequence of the suffering for the we-group
B^2	$10a\alpha$	bridge
D	$10a\beta-10b$	The consequence of the suffering for the addressee

 B^3 $$11a\alpha\beta$$ \$bridge\$ $$\bar{A}\bar{C}\bar{D}$$ $$11a\gamma{-}12$$ The consequence of the suffering for the many and the servant

This structure suggests a relationship between the we-group, the addressee of the poem, and the many. Verses $11a\gamma-12$ structurally correspond to verses 13-15, 4-9, and $10a\beta-10b$. Also thematically, they summarize these verses, stating the consequences of the servant's suffering for the servant and the many. Thus, it seems likely that the many comprise the many nations in verses 13-15, the we-group in verses 4-9, and the addressee in verse $10a\beta-10b$.

Chapter 3

The Ambiguous Identity of the Servant

Isaiah 53 frequently employs the rhetoric device of ambiguity. Section 3.1 lists the ambiguous expressions. The frequency of the rhetoric device of ambiguity suggests that it is employed intentionally, in order to create an ambiguous text. Specifically, verse $11a\gamma$ has several meanings which seem to contradict each other, with some stating that the servant is in need of reconciliation, while others state that the servant is righteous, and thus not in need of reconciliation. Since the ambiguity seems to be intended, this suggests that the identity of the servant is ambiguous. This is discussed in Section 3.2, suggesting that the servant seems to ambiguously designate both the servant Israel in need of reconciliation with God, and a righteous servant, who reconciles the servant Israel.

3.1 Ambiguities in Isaiah 53

יַשְׂבִּיל (13a): "he will succeed" and "he will understand." The former conforms with the parallel 52:13b, the latter conforms with 53:11aαβ.

מְשְׁחֵת (14aβ): construct state of both מְשְׁחָה "disfigurement" and מְשְׁחָת "anointment." Both are meaningful descriptions of the servant Israel. As a defeated and exiled people, the servant is disfigured. Further, since שֶׁבֶּן is missing, the servant Israel might be anointed with JHWH's spirit, like the speaker of Is 61:1:

But now hear, O Jacob my servant,
Israel whom I have chosen!
Thus says JHWH [...]
I will pour my spirit upon your descendants (Is 44:1-3)

(2aa): "on his own" and "before him [JHWH]." Both are meaningful descriptions of the servant Israel, who "grew up" during the forty years in the desert, isolated from other nations, and before the face of JHWH, which resided with Israel in the tabernacle (for example, Ex 27:21).

קמְסְתֵּר פְּנִים מְמֶּנוּ (3ba): "[he] hiding his face from us" and "[us] hiding our faces from him." Both are meaningful. The disfigured servant is hiding his face in shame, the onlookers turn their faces from him in disgust.

מוּסֶר שָּלוֹמנוּ (5ba) "God's punishment for our peace," "God's punishment we deserved as retaliation," and "God's chastisement for our peace." As has been shown in Section 2.4.1, can mean both "God's punishment for our peace" and "God's punishment we deserved as retaliation." The former is stronger, since peace most likely implies retaliation, while retaliation does not necessarily imply peace. Given the parallel "our healing," the former most likely is the primary meaning, thus, the latter is implied. Further, since מוּסֶר שָׁלוֹמנוּ, and being righteous implies to be in a right relationship with God, מוֹסֶר שָׁלוֹמנוּ can also mean "God's chastisement for our peace."

קבְּרָתוֹ (5bβ): "his slash," from תַּבּוּרָה, and "his being allied," passive participle from מוּסָר. The former conforms with the parallel מוּסָר, the latter might refer to the servant being made a בַּרִית עָם in Is 42:6.

מאָרֶץ חַיִּים (8ba) occurs ten times apart from Isaiah 53. In Jer 11:19, it means "life" as opposed to death, as made clear in Jer 11:21 ("those who seek your life"). This is probably also the main meaning in Ps 52:5. In Ezk 32:22–32 (six occurrences), it means the nation or land of Israel, as opposed to Assyria, Elam, Meshech and Tubal, and Egypt. This is probably also the main meaning in Ps 27:13. Concerning Isaiah 53, both meanings describe the exiled community. They are obviously cut off from the land of Israel, and the exile is described as "death" in Ezk 37. Similarly, both meanings might be intended in Ezk 26:20.

עַל (9ba): "although" and "because." "Although" relates to 9aa, "because" to 9aβ.

אַשָּׁשְּ (10a β): "liable by guilt" and "liability offering." The two meanings correspond to reconciliation by transfer of the burden of sin, discussed in Section 5.2, and reconciliation by atonement, discussed in Section 5.1.2.

דַּעָּת (11aβ): "insight" and "sweat." These correspond to the two meanings of הַּטָּר in 5bα. Insight is fostered by chastisement, and sweat caused by punishment. בְּרַשְׁתוּ can be taken with either verse 11aα or 11aγ. With verse 11aα, the resulting "he will be satisfied by his knowledge" parallels בְּרַשְׁתוֹ . יִרְאֵּה with verse 11aγ results in "by his knowledge, my righteous servant will show himself to be righteous to the many" or "by his sweat, a righteous will reconcile my servant before the many."

יַצְרֵיק עַבְרִי לְרַבִּים (11aγ): See Section 3.2 below.

3.2 The Ambiguous Identity of the Servant

As shown in Section 2.4.1, יַּצְּדִיק עַבְּדִי לְרַבִּים (11aγ) can have various meanings. Some of these interpretations seem to explicitly contradict each other, with some amounting to

the servant being in need of reconciliation, and others amounting to the servant being righteous, and thus not in need of reconciliation. Since the ambiguities in Isaiah 53 seem to be intended, it seems likely that also the various meanings of $11a\gamma$ are intended. The contradiction can be resolved if also the identity of the servant is ambiguous. The rhetoric device of ambiguity might indeed be frequently employed in order to emphasize that the identity of the servant is ambiguous.

The servant is Israel, "cut off from the land of the living" (8ba), thus most likely the exiled community. This conforms with שָּבֶּד designating Jacob/Israel in Isaiah 40–55. The servant being the exiled Israel conforms also with the description given of the servant in Isaiah 53. Conforming with Is 52:14, as a defeated and exiled people, the servant Israel is disfigured, but also anointed with JHWH's spirit (Is 44:1–3). Conforming with Is 53:2, the servant Israel "grew up" during the forty years in the desert, isolated from other nations, and before the face of JHWH. As exiled, the community had no majesty, was despised, had comparatively few members (lacking men, 3aa), was humiliated, oppressed, had been besieged (under arrest, 8aa), was condemned, and punished by God by being led into exile. The exile is likened to being dead in Ezekiel 37, which conforms both with the references to the servant's grave and death in verse 9a, and with the servant being likened to being led to slaughter (תַּבָּבָּן), since תַּבָּבָּן conveys both death and God's punishment.

The servant Israel is depicted as sinner in Is 42:24, 43:22–28, 44,21f, and 50:1. The we-group identifies themselves as אָלָשָׁב, which is commonly used for Israel (Waschke 1989, 864f). The we-group confesses that they went astray (אָלָשָׁב, 6aα). In Jer 50:6, the straying flock designates the exiled community. Thus, the we-group seems to be the exiled community, and thereby identical with the servant Israel. This is substantiated by the we-group confessing that they are sinners.

This depiction of the servant Israel as sinful in Is 42:24, 43:22–28, 44,21f, and 50:1 seems to contradict the description of the servant in Isaiah 53:9b as without violence and deceit. Mettinger (1983, 42f) suggests that the description of Israel in Is 51:7a ("Listen to me, knowers of righteousness, a people with my law in their hearts"), shows that the servant Israel is "simul justus et peccator." Another interpretation might be that "without violence and deceit" describes the exiled community as submitting (Is $53:7a\alpha$) to those who defeated them and to the situation in exile instead of revolting against it. This interpretation conforms with the exiled community being described as without violence and deceit, while it on the other hand can be called sinful for disobeying JHWH.

As sinful, the servant Israel is in need of reconciliation. Verse $11a\gamma$ states both that "a righteous reconciles my servant," and that "the righteous one, my servant, reconciles the many." The structure of Isaiah 53 suggests that the many comprise both the many nations, the we-group, and the addressee of the poem. Thus, the we-group, the servant Israel, is comprised by the many. Thus, the servant Israel is according to verse $11a\gamma$ reconciled by "a righteous" and "the righteous one, my servant." This suggests that "a

righteous" and "the righteous one, my servant" are identical. This is substantiated by the we-group declaring that they are reconciled by the servant being punished $(5b\alpha)$.

The righteous servant is likened to a שָּה, an individual animal of the flock באן. Thus, he seems to be one of the exiled community, and accordingly, everything that can be said about them, can be said about him. In addition, he seems to be the we-group's ally (5bβ).

The ambiguous identity of the servant is further supported by the debated verses Is 49:5f. Also in these verses, the rhetoric device of ambiguity presents an ambiguous servant:

```
5a\alpha
         And now JHWH says,
         who formed me in the womb to be his servant,
  β
                       Jacob to him,
         to return
  γ
         by returning
  δ
         such that Israel might be gathered to him,
6a\alpha
         he says: "It is too light a thing that you should be my servant
         to raise up
                            the tribes of Jacob
  β
         by [me] raising up
         and to return
                             the survivors of Israel;
  Υ
             [me] returning
```

The servant Israel is formed by JHWH by returning Jacob to himself. The righteous servant is called as JHWH's means to return the servant Israel to him.

As a consequence of the we-group being Israel, and "the many" comprising Israel, it seems likely that "the many" of Is 53:11f are the many nations of Is 52:15. In turn, it seems likely that the poem is addressed to every nation.

Chapter 4

Suffering in Isaiah 53

Isaiah 53 contains 37 descriptions of suffering, making suffering its dominant theme. The descriptions are the following:

- he was disfigured beyond resembling a human (14a β , 14b)
- he was isolated $(2a\alpha, 3a\alpha)$
- he was undernourished ("like a root from dry earth, he had neither appearance nor majesty", $2a\beta\gamma$)
- he was despised (גָבָוָה 3aα, 3bβ; לֹא חֲשַׁבְנָהוּ 3bβ)
- he suffered pain (אִישׁ מַכְאֹבוֹת 3aβ)
- he was sick (יִדוּעַ חֹלִי 3aβ; 10aα)
- he was ashamed (hiding his face from us, $3b\alpha$)
- we were sick $(4a\alpha)$
- we were in pain $(4a\beta)$
- he bore the burden of our sicknesses and our pains (מֵּכְאֹבֵינוּ סְבָלְם 4aα, מֵּכְאֹבֵינוּ סְבָלְם 4aβ)
- he was struck (4b β , 8b β)
- he was smitten by God $(4b\beta)$
- he was humiliated (4b β ; assigned a grave among the wicked, 9a α (Koch 1989, 1152); counted among felons, 12a δ)
- he was pierced $(5a\alpha)$
- he was crushed ($5a\beta$, $10a\alpha$)
- he was punished $(5b\alpha)$
- his slash $(5b\beta)$
- he bore the burden of $\sin (6b, 11b, 12b\alpha)$
- he was oppressed $(7a\alpha)$
- "he was led to slaughter" $(7a\beta)$
- he was arrested $(8a\alpha)$
- he was cut from the land of the living $(8b\alpha)$

- he died violently (9aβ)
- his soul was troubled $(11a\alpha)$
- his soul was stripped to death $(12a\gamma)$

4.1 Inclusive or Exclusive Suffering

As shown in Section 3.2, the we-group most likely is the exiled Israel, and the many are the many nations. The servant ambiguously refers to either the exiled Israel, that is, the we-group, or to a righteous servant who is a member of the exiled community. This section studies the relation between the suffering of the righteous servant and the suffering of the we-group and the many.

Hooker (1998, 96) defines inclusive suffering as a suffering alongside others, and exclusive suffering as a suffering instead of others. However, while the terms inclusive and exclusive are opposites of each other without further options, the servant might suffer neither along with nor instead of others. For example, Sweeney (2016, 215) identifies the servant with all Israelites who have suffered in the period between the Assyrian invasion in the eighth century up to the early Persian period. In this interpretation, whether correct or not, the servant Jacob/Israel hardly suffers along with or instead of other nations.

In the following, *inclusive* suffering presumes that the others suffer, and that the servant suffers like them or closely similar to them. *Exclusive* suffering refers to the servant suffering distinctively different from the others, who may suffer in a different way, or not at all. Specifically, exclusive suffering does not imply that the servant suffers instead of the others.

Of all descriptions of suffering in Isaiah 53, two describe suffering of the we-group: "our sicknesses" and "our pains" (4a). Sickness and pain are also extensively suffered by the servant: He is a "man of pains" and "known by sicknesses" (3a). Thus, the servant suffers inclusively. This is substantiated by the we-group likened to a flock (東北, 6aα), while the servant is likened to a $(7a\beta)$. Since the term $(7a\beta)$ is used for the individual animal within a $(7a\beta)$, the servant is likened to being one of them. Thus, he shares in their fate.

However, he not only shares the we-group's suffering, but he alone carries the burden of this suffering: "our sufferings he bore", and "our pains, he carried them" (4a). Thus, the servant seems to endure the suffering which he shares with the we-group more profoundly than the we-group does. Further, sicknesses and pains constitute only a small part of the suffering described in Isaiah 53. The vast majority of the descriptions refer to other types of suffering, and they refer only to the servant. His suffering is described in ways which by far exceed sickness and pain: he was led to slaughter, cut from the land of the living, died violently, his soul was stripped to death. "The nouns in this passage ('sorrows',

'sicknesses', 'wounds'), likewise the participles ('stricken', 'smitten', 'afflicted') [...] give a picture of a man subjected to every conceivable pain and indignity." (North 1964, 242)

4.1.1 Objections against the Servant Suffering Exclusively

Orlinsky (1977, 92) partitions Isaiah 53 into Is 52:13–15 and Is 53:1–12, and suggest that the former treats the servant Israel, and the latter the prophet Deuteroisaiah. Orlinsky (1977, 57) claims that the servant of Is 53:1–12 does not suffer in any extraordinary way. Rather, the prophet Deuteroisaiah suffers just like many other prophets. Orlinsky (1977, 60f) substantiates his claim by referring to Jer 11:19a,b as follows:

Concerning "like a sheep he was led to slaughter" (7aβ), Orlinsky (1977, 60) suggests that this merely asserts that the prophet unflinchingly endured a suffering which is typically endured by prophets. To substantiate that 7aβ describes an "occupational hazard" typical for prophets (57), Orlinsky (1977, 60) points to Jer 11:19a, "but I was like a gentle lamb led to the slaughter." However, Jer 11:19a uses a different word for lamb, and the metaphors are different. In Isaiah 53, the metaphor indeed is about the servant unflinchingly enduring the punishment. As made explicit by Jer 11:19bα, "And I did not know it was against me that they devised schemes," Jeremiah 11:19a is about the prophet's naivety, his complete unawareness of his opponents' murderous conspiracy. Thus, Is 53:7 cannot be likened to Jer 11:19. Furthermore, while verse 7aβ emphasizes that the servant endures his suffering unflinchingly, this does not diminish the extend of the suffering described in Isaiah 53.

Concerning "he was cut from the land of the living" (8b β), Orlinsky (1977, 60f) refers to Jer 11:19b β "let us cut him off from the land of the living," which is uttered by Jeremiah's opponents planning to kill him. Orlinsky (1977, 61) concludes that since Jeremiah lived after this, Is 53:8b β cannot mean that the servant died. However, Jer 11:19b β does not state an event which has happened, but only the content of the opponents' plan: "they devised schemes, saying" (Jer 11:19b α). Since God kills everybody plotting to murder Jeremiah (Jer 11:21–23), the opponents can never execute their murderous plan, and consequently, Jeremiah lives despite the opponents planning to cut him off from the land of the living. In contrast, Is 53:8b does not describe someone merely planning to kill the servant, but states that the servant was actually cut from the land of the living. Therefore, it is impossible to conclude from Jer 11:19 that Is 53:8b β does not mean that the servant died. And even if Orlinsky (1977, 61f) is correct in claiming that the servant did not die, this does not imply that the servant did not suffer profoundly.

That the servant does not suffer in an extraordinary way, might further be argued based on the description of his suffering resembling descriptions of suffering in individual complaint psalms, for example Psalm 22. However, there is a fundamental difference between Isaiah 53 and individual complaint psalms. The individual complain psalms explain

suffering as the "devastating consequence of the absence of YHWH" (Lindström 1994, 96). As soon as JHWH abandons a person, the person becomes defenseless and is immediately attacked by evil forces (95f). Contrary to this, Isaiah 53 depicts a crisis not caused by the absence of JHWH, but a crisis which is caused by JHWH and willed by JHWH. The only suffering in the Old Testament which potentially can be compared to that of the servant is the suffering endured by Job, who certainly suffers in an extraordinary way. But even he is not crushed by God ("O that it would please God to crush me (יִדְּכָאָנִי)," Job 6:9). Also, God is not actively inflicting Job's suffering, but tolerating Satan's infliction. In Isaiah 53, it is God who actively makes the servant suffer, thus the servant's suffering surpasses even Job's suffering.

4.1.2 The Servant's Exclusion

That the servant suffers exclusively, is further supported by Isaiah 53 throughout the text emphatically isolating the servant from the others (the we-group and the many). This is done in several ways:

He is explicitly stated to be isolated $(2a\alpha)$ and deserted $(3a\alpha)$.

He is set apart from the others by emphatic הוא opposed to אָּבְּדְעָה (4), and opposed to "our diseases" (4a α), "our felonies" (5a α), "their sins" (11b), and "the offence of many" (12b). Several manuscripts, the Syriac translation, and the Vulgate also have an extra in contrast to "our pains" (4a β).

Isaiah 53 gives only few characteristics of the servant and the others. These few characteristics are thus emphasized. The servant's righteousness is emphasized by describing him as without violence nor deceit (9b), and by the use of מון in two different meanings in verse 8, which highlights the unexpectedness of sentencing the servant to punishment. The others, on the contrary, are described using only one characteristic, namely their sinfulness:

- our felonies $(5a\alpha)$
- our sins $(5a\beta)$
- we all went astray (6aα): "This [...] is a straying from God" (Koole 1998, 297), where "מעה [...] ein schuldhaftes Abirren vom rechten Weg impliziert. Das Verb meint in theologischer Verwendung durchweg ein sündiges, bewusst gewähltes Verhalten, das dem Willen JHWHs widerspricht" (Berges 2015, 255)
- each of us turned to his own way (6aβ): "פנה" [...] implies an intentional breaking away from Yahweh" (Koole 1998, 297). דֶּרֶב is the opposite of דֶּרֶב, "the conduct required by God," thus going on one's own way implies not conducting oneself according to God's requirements.
- the sin of us all $(6b\beta)$
- 1. The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "דֶּרֶךְ 5.b"

- the felony of my people (8b β)
- their sins (11b)
- the felons $(12a\delta, 12b\beta)$
- the offence of many $(12b\alpha)$

The contrast between the servant and the we-group is further emphasized by the contrasting attributes in verse 5b (our peace versus God's punishment on him, his slash versus our healing) and in verse 8b β (the felony of my people versus his strike), and by the sheep metaphor: While the servant is compared to sheep in a positive characteristic, submissiveness (7a α), the we-group is compared to sheep in a negative characteristic, their tendency to stray their own way and get lost (6a α) (Oswalt 1997, 390f).

Finally, apart from the sheep metaphor in verse 6, Isaiah 53 contains four verbs in first person plural. All of them have the servant as object, thus explicitly excluding him from the we-group: וְנְרְאֵהוּ, that we should look at him (2b), וְנְרְאֵהוּ, that we should desire him (2b), הַשַּׁבְּהָהוּ, we valued/evaluated him (3bβ, 4bα). Likewise, the servant is excluded from the many in verse 14aα, since they are appalled at him.

4.2 Conclusion

While the servant shares in the suffering of the we-group and thus suffers inclusively, the servant's suffering by far exceeds the others' suffering in quality and in the types of suffering endured. Thus, he suffers exclusively. Isaiah 53 emphasizes the servant's exclusive suffering, both by the quantity of expressions describing it, and by emphatically isolating the servant from the others.

Chapter 5

The Consequence of the Servant's Suffering

This chapter studies the relation between the servant's suffering and reconciliation of the we-group and the many with God.

The relationship with God of both the we-group and the many was broken due to their felonies (אַשָּענה (5aα), פָּשָׁע עַמִּי (8aβ) פּֿשָׁע (12aδ, 12bβ)). Because of the servant's punishment, the we-group is reconciled with God: God's punishment for our peace with God was on him, by (instrumental ב) his slash we were healed (5b). Also the many are reconciled by God's righteous servant enduring the burden of their sin (11aγb).

5.1 Common Interpretations of the Servant's Reconciling Ministry

The interpretations reviewed in Chapter 1 suggest that the servant achieves reconciliation either by his message fostering contrition or faith, or by the his suffering being atoning.

5.1.1 Reconciliation by Message

Janowski (1993) and Hermisson (2017b) interpret the righteous servant as the prophet Deuteroisaiah, who by his message fosters contrition or faith, and thus returns Israel to JHWH. They seem to interpret reconciliation as comprising two stages. As a prerequisite for reconciliation, the servant dies as compensation for Israel's guilt, while Israel ultimately is reconciled by its contrition or faith.

Concerning the first stage, Janowski (1993, 19) interprets Israel as guilty to a degree which made it impossible for Israel to compensate for its obligation. In order to have a future, Israel had to be released from its obligation. It is released by the servant dying as "Schuldtilgung" (

Similarly, Hermisson (2017b, 398) interprets the death of the

servant as "vollgültige Sühneleistung" (קּשָׁבֶּּ), a merit which fully compensates for the guilt of the servant's people. "Hier aber geht es nicht um die Aufhebung der Schuldfolgen, sondern um ihre Übertragung auf einen Unschuldigen, so, dass die Täter schuld- und straffrei werden" (Hermisson 2017b, 369).

Concerning the second stage, Janowski (1993, 13,20) suggests that the servant's death together with the message of Is 52:13–53:1 causes Israel to understand the meaning of the servant's death, and by this, its own situation of being guilty. Only the confessing, contrite Israel is reconciled with God (20). Similarly, Hermisson (2017b, 424) suggests that the servant's resurrection causes Israel to believe in the servant's message of JHWH's plan for salvation. Since their guilt was their unbelief in his message, and they now believe it, they are reconciled with God.

However, this second stage seems inconsistent with the interpretation of אָשָׁבּי "Schuldtilgung," or "vollgültige Sühneleistung für die Schuld," which frees the others from their guilt and punishment. If they are freed from their guilt and punishment, then they are reconciled with God. Further, proposing reconciliation by contrition or faith seems contrary to Is 53:5, which explicitly states that peace and healing are effected by the servant's punishment, not by his message, nor by the others' contrition, confession, or faith. While the we-group's insight into their being guilty and their confession are reported in Isaiah 53, these are not what reconciles them. Rather, these are their reaction to being reconciled by the servant's suffering (see Section 5.2), or their reaction to being chastised (see Section 5.3). Finally, Isaiah 53 does not mention any message of the servant. "What we heard" (1a) is not necessarily a message by the servant. Hermisson (2017b, 355) interprets the content of "what we heard" as the two speeches by JHWH which frame Isaiah 53. However, because of עבּרִי in both parts of this frame, they can hardly be proclaimed by the servant. Hermisson (2017b, 418) interprets "each to his own way we turned" (6a) as the we-group not paying heed to the prophet's message. However, בּרָכּוֹ is the opposite of יהוה, "the conduct required by God," which refers to God's will in general, not to a specific revelation.

uith his earlier works, he regards Israel as guilty to a degree which made it impossible for Israel to compensate for its obligation. Thus, Israel had to be released from its obligation in order to have a future (58). This happens by the servant dying as "Schuldtilgung" (写文本) (54f). The message of Is 52:13-53:1 causes Israel to reflect and to understand that the servant suffered not because of his, but because of their deeds (55f). Janowski (2009, 56) calls this insight of being guilty "the beginning of transformation," followed by the ambiguous "Die Wirklichkeit der Stellvertretung erschließt sich den Wir aber nicht einfach durch Reflexion oder durch Entschluß, sondern durch das im Bekenntnis von Jes 53,4 ergriffene Wort, das JHWH nach Jes 52,13-15 über den Erfolg seines Knechts spricht."

1. The Hebrew and Aramaic Lexicon of the Old Testament, s.v. "דֶּרֶךְ 5.b"

This can mean that by confessing, the we-group understands the reality of substitution, but can also mean that the we-group gets access to the reality of substitution by their confession. The latter seems to be what is intended by Janowski, first, because "get access to" is the primary meaning of "erschließen," and second, "by their confession" conforms better with the latter than with the former. Thus, insight and confession are necessary requirements for being substituted, and by that, the necessary requirements for being reconciled.

Janowski (2009) follows this up with statements which are new with respect to his earlier works: The suffering and death of the servant constitute a "Stellvertretung für die Sünder" (57, his emphasis). This substitutionary suffering is interpreted as salvation, "der von JHWH bestimmte Weg zum Heil" (57, his emphasis). The message of the servant is his testimony for God's will to be the savior of the world (57).

Altogether, the conception of reconciliation seems to be the same as in Janowski's earlier works, having the same inconsistency. The servant dies as a substitute, assuming the consequences of the others' deeds. However, this is only a prerequisite to reconciliation, for which insight and confession of guilt seem to be decisive.

5.1.2 Reconciliation by Atonement

The servant's suffering has been interpreted as cultic expiation, both as the servant acting as scapegoat, as liability offering, or as cultmetaphorical compensation.

The Servant as Scapegoat

Mettinger (1983, 41) suggests that the exiled community achieves reconciliation by acting as Israel's scapegoat. He substantiates this as follows: אָרֶץ הַּיִּים should be interpreted as the opposite of desert, thus, Isaiah 53:8 describes the servant as being driven into the desert, just like the scapegoat on the Day of Atonement. Further, the scapegoat's carrying away of the sins is expressed by נשא מון, which Mettinger (1983, 41), following Zimmerli (1969, 236–244), finds alluded to by the expressions סבל מון (Is 53:1), and מבל מון (Is 53:12).

Mettinger's arguments are not convincing. First, it is the sinful servant Israel who in Isaiah 53 is in need of reconciliation. The exiled can hardly act as their own scapegoat. For the interpretation of אֶּבֶץ הַיִּים as the opposite of desert, Mettinger (1983, 41) refers to Gerleman (1980, 41), who in turn states without further argumentation that in the Old Testament, אֶּבֶץ הַיִּים is not opposed to death, but to an uninhabited place, specifically the desert. Gerleman (1980, 42) also states: "Häufig ist offenbar Israels Land gemeint." As shown on page 44, אֵבֶץ הַיִּים is never opposed to the desert, but either to the land of

^{2.} Duden, s.v. "erschließen," accessed December 12, 2017, https://www.duden.de/rechtschreibung/erschlieszen

Israel, or to death. Thus, it cannot be concluded from 53:8 that Israel is driven into the desert like the scapegoat. Further, as shown in Section 2.2.1, משֹא חמא and משֹא in Isaiah 53 express enduring the consequences of sin. They cannot be equated to the cultic elimination of sin expressed by נשׂא עון.

However, Isaiah 53 does allude to the Day of Atonement as follows:

- in Is 53:8, the servant is taken (לֶּקֶּם) and subsequently cut (נְגוַר) from the land of the living (מֵאֶּרֶץ חַיִּים)
 - קמו is used for the priest taking the animals to be used as sin offering and burnt offering for Israel (Lev 16:5), and for taking the two goats in order to present them to JHWH (Lev 16:7).
 - the scapegoat is sent to "a land of cut off" (אֶרֶץ נְּזֶרָה, Lev 16:22)
- while the goats in Lev 16 are designated by טָּה, a שָּׁמָר, which the servant is likened to in Is 53:7, can be a goat
- the priest sprinkles (מה hiphil) the blood of both his and Israel's sin offering four times (Lev 16:14–19)
- Israel's אָוֹן and אָוֹן are confessed by the high priest over the head of the scapegoat (Lev 16:21).
- the servant is humiliated (מְּעָנֶה, pual, Is 53:4aβ). On the Day of Atonement, to be humiliated (שנה pual, Lev 16:29) and humiliating oneself (שנה piel, Lev 16:29, 31, 23:27, 32) is required.

The Servant as Liability Offering

Sweeney (2016, 209f, 213) interprets the servant's suffering as atoning sacrifice based on

- interpreting יוֶה (15aa) as cultic "he will sprinkle"
- interpreting אַשֶּׁבּ (10aβ) as cultic "guilt offering"
- the servant being led to slaughter like sheep (מָּב, 7aβ)

As shown in Section 2.2.2, it is unlikely that אָדָ means "he will sprinkle." The sheep's slaughter in 53:7 most likely is profane rather than atoning, since 53:7 does not use expressions for cultic slaughter, אַדָּם and שַּבּח but אַדָּם, which denotes secular slaughter (see Section 2.2.2). That the slaughter is secular rather than atoning, is substantiated by the parallel comparison "like a sheep before its shearers", since shearing is profane, not sacrificial treatment of sheep (Berges 2015, 260). Deuteronomy 17:1 emphasizes that the servant, disfigured and sick, is disqualified as a sacrificial animal: "You must not sacrifice to the Lord your God an ox or a sheep (אַדֶּה, which the servant is likened to in 7aβ) that has a defect, anything seriously wrong; for that is abhorrent to the Lord your God." Further, in the Old Testament, human blood cannot atone. Finally, the focus of Isaiah 53 is on the servant's suffering. Sacrificial animals die, but they are not made suffer.

However, there are several allusions to the liability offering, especially the expression אָשָׁבּבּ, and also אוֹם hiphil. Furthermore, the servant is described as without deceit and violence. While he is physically disfigured and sick, and thus disqualified as liability offering, it can be argued that he is morally blameless, and thus metaphorically qualified.

The Servant as Cultmetaphorical Compensation

Berges (2015, 269) interprets "", and thereby the servant, based on 1.Sam 5f, as a "cultmetaphorical" compensation. Just as the Philistines in 1.Sam 5f sent a representation of their suffering as "", so the Israelites can set a representation of their suffering (being exiled and disregarded), that is, their suffering and disregarded representative, as theirs.

Berges' interpretation requires that the servant is the Israelites' representative. Isaiah 53 does not state that the servant is their representative. Since he is despised, deserted, and not valued by them, this seems unlikely. If he is not their representative, then Berges' interpretation of at the representation of the we-group's suffering is impossible.

But even if he is their representative, Berges neglects that while the Philistines send a representation of their suffering, this representation is a compensation, a valuable gift, "five gold tumors and five gold mice" (1Sam 6:4). The focus of the philipping is on the value, making it a punishment for the Philistines to send it. On the contrary, it is not a punishment for the Israelites to send someone whom they do not value. It is not possible to set a disregarded person as an per

5.2 Reconciliation by Transfer of the Burden of Sin

Isaiah 53 states that the we-group is guilty of felonies $(5a\alpha, 8b\beta)$, and sins $(5a\beta, 6b\beta)$. The righteous servant, on the contrary, is innocent $(9b, 11a\gamma)$. Nevertheless, the servant is sentenced $(8a\alpha)$, and the servant is punished by and according to the will of God $(5b\alpha, 6b, 10a\alpha)$. The we-group, on the contrary, is never stated to be sentenced or punished. For each instances of their felonies and sins, it is the servant who is punished because of them $(5a\alpha, 5a\beta, 6b, 8b)$, while Isaiah 53 never states that the we-group is punished. The punishment of the servant is the punishment the we-group deserves for its sin, "the burden of the sin of us all" $(6b\beta)$. The consequence of the servant's punishment is the same as if the we-group was punished: They are reconciled with God (5b). Verse 5b emphasizes that they are reconciled with God because of the servant being punished.

Similarly, Isaiah 53 states that the many are felons ($12a\delta$, $12b\beta$), guilty of sins (11b) and and offences ($12b\alpha$). Isaiah 53 never states that the many are sentenced or punished. That it is the servant who carries the burden of their sins (11b, $12b\alpha$), is emphasized by in both verses. Again, the consequence of the servant's punishment is the same as if

the many were punished. They are reconciled with God (11a γ). Verse 11a γ emphasizes that the many are reconciled by the servant.

The simplest explanation for the punishment of the servant would be that he himself is guilty and thus justly punished. However, the servant is explicitly stated to be innocent, and he is emphatically isolated from the others with respect to his righteousness in contrast to their sinfulness. While Isaiah 53 explicitly states that he shares in their sicknesses and pains, Isaiah 53 does not give him a share in their offences, felonies, and sins. The servant is innocent.

Verse 5a ("he was pierced because of our felonies, crushed because of our sins") might indicate that the servant suffered as the victim of the we-group's felonies. However, this does not explain why the servant suffers according to God's will, and why the others are reconciled by his suffering.

Another explanation would be that the servant is the victim of a miscarriage of justice. Again, this does not explain why the innocent servant is punished in accordance with the will of God.

The only scenario in which the innocent servant can justly assume the guilt of the others' sin and endure the punishment that they deserve, is that the servant is *forensically liable* for their sin. This explains that the punishment the servant endures, is the punishment deserved by the we-group. This explains why he has to carry the burden of the many's sin. Having someone who is forensically liable for them and who is punished instead of them, explains that the others are justly neither sentenced nor punished, and nevertheless reconciled with God as if they themselves had been punished.

That the servant is liable for the others is substantiated by verse 10, which guarantees the addressee of the poem that JHWH's will succeeds if the addressee appoints the servant as liable for his guilt. According to $10a\alpha$, the will of JHWH comprises crushing the servant. According to $5a\beta$, the servant is crushed because of the sins of the we-group. Thus, verse $10a\beta$ guarantees to the addressee that if the appoints the servant to be liable for him, then the servant will be crushed for the addressee's sin. The addressee, in turn, is released from punishment. Thus, the servant will see the addressee as his offspring and prolong the addressee's days $(10a\gamma)$.

5.2.1 Objections Against Transfer of the Burden of Sin

Whybray (1978, 60f) argues that verse 6b "certainly does not itself imply that their punishment was transferred to him". While verse 6b itself actually implies just that (see Section 2.4.1), in any case verse 6b has to be read in the context of Isaiah 53: The innocence of the servant, him being sentenced and punished according to God's will, and that while Isaiah 53 does not state that the others are punished or had their sin expiated in any other way, they nevertheless are reconciled with God as if they had been punished,

or expiation had happened. Isaiah 53:6b describes a transfer of punishment from the we-group to the servant.

Concerning סבל עון (11b), Whybray (1978, 29f) refers to the only other occurrence in Lam 5:7 and points out that there is "no question here of vicarious suffering: there is cause and effect, but not substitution", with the cause and effect being the common fate of sin, guilt, and punishment belonging together and binding together ancestors and descendants, who themselves confess to be sinners (Lam 5:16). However, the descendants themselves being sinners distinguishes them from the servant, who is righteous. His bearing of the burden of the many's sins can therefore not be denied to be substitutionary on the basis of Lamentations 5.

While Hägglund (2008, 67) agrees that "YHWH is the cause of the suffering of an innocent servant, of a suffering that the 'we' have deserved," he argues that this not necessarily implies that the we-group is "freed from their guilt. Rather, they have escaped their own suffering. Nothing is said about atoning suffering" (56). Hägglund is correct that Isaiah 53 does not describe the servant's suffering as atoning, and that the servant's suffering does not necessarily imply that the we-group is freed from their guilt. However, the suffering endured by the servant is a specific suffering. It is the suffering that the we-group deserves because of their sin (5ba, 6b). Thus, the we-group does not simply "escape their own suffering". Rather, the punishment they deserve is transferred to the servant. This must include a transfer of their guilt, first, because sin, the guilt incurred by sin, and the punishment for the guilt form a unity which cannot be broken up, and second, because the servant cannot be punished according to the will of God if he is not guilty. Since he himself is innocent, the guilt he is punished for is the guilt that is transferred to him from the others. Finally, Isaiah 53 states that the others are reconciled with God. This could not be if they still were guilty. Thus, the others are freed from their guilt.

Isaiah 40:1f

"Comfort, comfort my people, says your God. Tell the heart of Jerusalem [...] that her sin (שָּוֹן) is paid, that she has received from the Lord's hand double for all her sins (אַנוֹן)." (Is 40:1f)

Referring to Is 40:1f, it is argued that the servant cannot vicariously suffer Israel's punishment, since Is 40:1f states that "Israel had already experienced that punishment [...] and had thereby fully expiated her sinfulness (40.1-2)" (Orlinsky 1977, 58). However, it can be questioned whether Is 40:1f has this implication for Isaiah 53. The one who in Is 40:2 is said to have suffered double for all her sins is Jerusalem, not Israel. Isaiah 40-55 addresses the exiled community, who are called to return to the land of Israel. Isaiah 40:1 calls the people in exile to comfort (second person plural) Jerusalem. Thus, the "my people" of Jerusalem (Is 40:1f) and of Is 53:8 are most likely different groups. This is substantiated by "my" in Is 40:2 referring to God, which "my" probably does not in

Is 53:8. The people of Jerusalem are the people left there by the Babylonians, while those of Is 53:8 are the Israelites in Babylonian exile. Thus, it cannot be concluded from Is 40:1f that the we-group in Isaiah 53 has endured the punishment it deserved.

Isaiah 43:25 and 44:22

In both verses, God tells Jacob/Israel that he himself annihilates their felonies and offences. Thus, it can be argued that the suffering of the servant cannot reconcile, since the others' sins are already expiated by God. However, both verses do not explain how Israel's sins are annihilated. Isaiah 53 gives this explanation.

Ezekiel 14

Ezekiel 14:12-20 might give the impression that not even the most righteous person can save additional persons. However, this passage counters Gen 18:23–19:16 and Jer 5:1, where God concedes to saving cities because of the mere presence of righteous persons. Ezekiel 14 declares that God is not making any such concessions anymore. Even if these three men were in it [the land] (בְּתוֹכָה, Ezk 14:14,16,18,20), nobody else would be saved.

The passage excludes the possibility of a righteous person saving someone else *along* with himself by his mere presence. It does not exclude the possibility of saving someone by suffering their punishment instead of them.

Ezekiel 18

Ezekiel 18:1–20 might give the impression that guilt cannot be transferred, not even from a father to his son or the other way round. However, in Ezk 18:1–20, God is answering the Israelites' accusation that he is making them suffer for their ancestors' sins (Ezk 18:2). God declares that this accusation is unjustified: He will neither punish a son for his father's sin nor the other way round (Ezk 18:3f). God rejects that he would transfer the punishment from one person to another just because they are related. This does not exclude transfer of guilt and punishment from a person to another if the other person is liable for the first.

The passage substantiates that transfer of punishment from one person to another was possible in Old Testament thought. If such transfer was unthinkable for them, the Israelites could not suspect God of transferring their ancestors' punishment to them.

5.3 Reconciliation by Chastisement and Insight

As shown in Section 2.3, Isaiah 53 contains a significant number of expressions related to knowledge of God and keeping his commandments.

Verse 5b\(\alpha\) can be interpreted as "God's chastisement for our peace was on him." When the source of chastisement is JHWH, chastisement is redemptive by restoring the chastised to proper conduct (Branson 1982, 695). The exile may be interpreted as chastisement from God. Consequently, the chastised servant is the servant Israel.

That the servant Israel learns from this chastisement, is substantiated by the proclamation that the servant will understand (13a, 11aαβ). The expressions used convey experiencing God through his actions, an experience which leads both to appropriate faith in and fear of JHWH, and to analyzing one's own conduct, resulting in adapting one's conduct to JHWH's requirements.

As the Masoretic verse partition indicates by making "by his knowledge" (בְּרַשְׁתוֹּ) its own stich (11aβ), "by his knowledge" can not only be taken together with 11aα, but also with 11aγ. This is explicitly done in 1QIs^a by inserting 1 before בְּרַשְׁתוֹּ. Verse 11aβγ can be interpreted as "by his knowledge, my righteous servant will prove himself righteous before the many." הַּבְּרַשְׁתוֹּ means an acquaintance with God which constitutes itself in righteous conduct. Because of the lesson learned by God's chastisement, the servant Israel can lead a righteous life. By leading a righteous life, he both is righteous and appears righteous before the many. By leading a righteous life, that is, adhering to the conduct required by God, he also is in peace with God.

Leading a righteous life comprises insight. However, this insight is not only insight into one's own situation, as Janowski (1993, 20) interprets, but rather appropriate fear of JHWH. Such insight does not mainly aim at contrition, but at adapting one's conduct to JHWH's requirements. It is righteous conduct, not contrition, which constitutes a righteous life. Similarly, experiencing God leads to faith (Is 41:20), but it also leads to analyzing and adapting one's conduct. The aim of chastisement is not contrition of faith, but adaption of conduct to the requirements (Sæbø 1971, 739). Contrary to the suggestions by Janowski (1993) and Hermisson (2017b), being righteous comprises not only contrition and faith in God, but also corresponding conduct (Johnson 1989, 921).

It seems that Janowski (1993) and Hermisson (2017b) do not fully acknowledge the ambiguity of Isaiah 53 concerning the identity of the servant and the corresponding means of reconciliation. The righteous servant suffers the consequences of the others' sin, which frees the others from their guilt and punishment. Contrary to the suggestion of Janowski (1993) and Hermisson (2017b), it is not primarily this suffering which leads to insight. It is primarily the chastisement of the servant Israel in exile which leads to insight and correspondingly adapted conduct.

5.4 Conclusion

Isaiah 53 is ambiguous concerning how peace with God is achieved. It states both reconciliation by chastisement and insight, and by transfer of the burden of sin, and clearly

alludes to various means of cultic expiation of sin.

Concerning chastisement and insight, Isaiah 53 states that the servant Israel is chastised and thus restored to the conduct required by God. However, a main concern in Isaiah 53 and Isaiah 40–55 is Israel's sin. Conducting a righteous life does not expiate sin (Johnson 1989, 921). Thus, while adhering to the conduct required by God maintains peace with God, sin still has to be dealt with.

Concerning transfer of the burden of sin, Isaiah 53 states that the we-group's and the others' burden of sin, that is, the guilt incurred by it and the punishment for it, is transferred to the righteous servant. The others are thus freed from their guilt, and thereby reconciled with God. The others' burden of sin is transferred to the servant by God, thus, the transfer must be justified. For this transfer to be justified, the servant must be forensically liable for the others.

Concerning cultic expiation, Isaiah 53 alludes to both the Day of Atonement and to the liability offering. However, the servant is not decisively depicted as a liability offering or the scapegoat. Further, for an offering or the scapegoat to be atoning, sin has to be confessed before the animal is slaughtered or sent away, respectively. In Isaiah 53, the confession in verses 5f probably occurs after the servant has been taken away. In addition, Isaiah 53 lacks the central term for atonement, and does not mention the atoning blood. Against this, it might be argued that if was used in Isaiah 53, or if blood was mentioned, then the cultic aspects of the text would be so dominant that the forensic aspects might disappear in comparison.

That Isaiah 53 regards reconciliation by forensic transfer of the burden of sin rather than by cultic atonement, is substantiated by cultic atonement not being a theme in Isaiah 40–55, whereas God's judgment and punishment is a common theme in these chapters:

- 41:1, God summons the nations to assemble for a court trial
- 41:21, JHWH's call to present evidence in favour of the idols, which are subsequently judged to be a delusion (41:29)
- 42:24f, JHWH's punishment for Israel's sin
- 43:9-12, JHWH's call to the nations to witness for the idols, and to his servant to witness for him
- 43:26-28, God's call to Israel to go to trial against him, his judgment over their ancestors and interpreters and resulting punishment
- 44:8f, JHWH's call to Israel to be his witness, and judgment over the idols and their witnesses
- 45:20f, God summons the nations to assemble for a court trial and to present the idols' case
- 50:1, JHWH's punishment of sin
- 50:8f, the servant's declaration that no adversary can contend with him and condemn

him

Overall, concerning the problem of sin, reconciliation by transfer of the burden of sin seems to be more decisively stated than reconciliation by atonement. Nevertheless, both the Day of Atonement and the liability offering are clearly alluded to. This may suggest that reconciliation by transfer of the burden of sin happens within a framework of cultic atonement, and thus is atoning. The burden of sin is not only transferred, but by this transfer, the others' sin seems to be completely annihilated (Is 43:25, 44:22). Consequently, the others are completely reconciled.

Chapter 6

Vicarious Substitution in the Old Testament

Transfer of punishment is traditionally designated by *vicarious suffering*. Both opponents and proponents of the understanding of the righteous servant's suffering as vicarious remark that vicarious suffering is without parallel in the Old Testament, for example Orlinsky (1977, 54) and Spieckermann (2004, 1).

Spieckermann (2001, 126) argues that whereas vicarious substitution in the sense of representation is a common and widespread concept in the Old Testament, "ist die Stellvertretung im Sinne der Substitution sehr selten belegt. Freilich gewinnt sie erst in diesem Verständnis theologische Prägnanz. Unter dem Aspekt der Substitution rückt die stellvertretende Lebenshingabe eines Menschen zum Heil anderer ins Zentrum theologischer Reflexion. Dieser Gedanke ist nur in einem alttestamentlichen Text, dem vierten Gottesknechtslied in Jes 52,13–53,12, realisiert worden."

Spieckermann narrows the concept of vicarious substitution down to vicarious suffering by arguing that only vicarious dying is theologically relevant. Consequently, Spieckermann does not find this concept in any other text than Isaiah 53.

However, even if vicarious suffering is the only theologically relevant instance of vicarious substitution, the concept of vicarious suffering might still be inspired by the more general concept of vicarious substitution, with vicarious suffering as a specific application of the concept of vicarious substitution. The concept of vicarious substitution is found in several texts in the Old Testament.

The Sacrificial Animal

The sacrificial animal substitutes for the sinner and dies instead of him (Janowski 1982, 220).

The Levirate

If a husband has failed to continue his name by not fathering a son before he dies, the dead husband's brother is obliged to father a son in his dead brother's name (Deut 25:5f). Ruth 4:5f also illustrates that it is possible to transfer this obligation.

- Gen 38:6–26 Then Judah said to Onan, "Go in to your brother's wife and perform the duty of a brother-in-law to her; raise up offspring for your brother." (Gen 38:8)
- Deut 25:5–10 When brothers reside together, and one of them dies and has no son, the wife of the deceased shall not be married outside the family to a stranger. Her husband's brother shall go in to her, taking her in marriage, and performing the duty of a husband's brother to her, and the firstborn whom she bears shall succeed to the name of the deceased brother, so that his name may not be blotted out of Israel. (Deut 25:5f)
- Ruth 4:3–16 Then Boaz said, "The day you acquire the field from the hand of Naomi, you are also acquiring Ruth the Moabite, the widow of the dead man, to raise up the dead man's name on his inheritance." At this, the next-of-kin said, "I cannot redeem it for myself without damaging my own inheritance. Take my right of redemption yourself, for I cannot redeem it." (Ruth 4:5f)

The Levites as the Firstborns' Substitutes

The Levites are appointed by God to be the firstborns' substitutes:

- Num 3:12 I hereby accept the Levites from among the Israelites as substitutes for all the firstborn that open the womb among the Israelites. The Levites shall be mine.
- Num 3:41 But you shall accept the Levites for me I am the Lord as substitutes for all the firstborn among the Israelites, and the livestock of the Levites as substitutes for all the firstborn among the livestock of the Israelites.
- Num 3:45 Accept the Levites as substitutes for all the firstborn among the Israelites, and the livestock of the Levites as substitutes for their livestock; and the Levites shall be mine. I am the Lord.
- Num 8:14,16 Thus you shall separate the Levites from among the other Israelites, and the Levites shall be mine. For they are unreservedly given to me from among the Israelites; I have taken them for myself, in place of all that open the womb,
- Num 8:18 but I have taken the Levites in place of all the firstborn among the Israelites.

Spieckermann (2001, 136) accepts the conclusion by Körting (1999, 183) that "die Idee der Substitution [...] in bezug auf Num 8,10 nicht haltbar. Die Leviten werden Jhwh als הנופה der Gemeinde übergeben. Die Gemeinde gibt etwas von sich weg, jedoch nicht sich selbst." This argument neglects that while the congregation indeed gives of itself, it was supposed to give its firstborns: "for all the firstborn are mine; when I killed all the firstborn in the land of Egypt, I consecrated for my own all the firstborn in Israel, both human and animal; they shall be mine. I am the Lord" (Num 3:13). However, God accepts the Levites as the firstborns' substitutes.

Judah's Appeal to be Vicariously Punished

The maybe closest parallel to Isaiah 53 is found in Gen 44. Benjamin has allegedly stolen Joseph's silver cup, and Joseph has sentenced him to become his slave. Judah appeals to Joseph to let him endure the punishment instead of Benjamin: "Now therefore, please let your servant remain as a slave to my lord in place of the boy; and let the boy go back with his brothers." (Gen 44:33) "Juda bittet, anstelle Benjamins die Strafe der Versklavung tragen zu dürfen" (von Rad 1967, 345). Doing so, Judah does not only intend to save Benjamin, but also his father, who would die if Benjamin stays in Egypt (Gen 44:31). It can be argued that Judah offers to vicariously endure Benjamin's punishment in order to save his father's life.

6.1 Conclusion

Vicarious suffering might be inspired by applying the more general concept of vicarious substitution to suffering. Vicarious substitution is found in several Old Testament texts. Since "sin was a dominant concern [...] forgiveness of sin formed a vital doctrine in Israelite faith" (Cover 1992, 31, 39), and both having and keeping one's firstborn was of vital importance in a culture where people depended on having sons, vicarious substitution in the forms of the sacrificial animal, the levirate, and the Levites substituting for the firstborns, was a vital, well-established, and familiar concept. That it can be applied to being vicariously punished, is demonstrated by Judah requesting to vicariously endure Benjamin's punishment.

Chapter 7

Conclusions

This thesis studies the relation between the servant's suffering and the others' reconciliation as it is presented in Isaiah 53. Current interpretations of Isaiah 53 present two suggestions for this relation. Reconciliation is suggested to be by the servant's message fostering contrition or faith, or by his suffering being atoning.

The interpretation of the servant fostering contrition or faith seems strongly dependent on identifying the servant as a prophet. This is substantiated by that those who interpret the servant's suffering to be atoning, do not identify the servant with a prophet, but with Israel.

The identification of the servant depends on what is assumed to be the original context of Isaiah 53. If the context of Isaiah 53 is Isaiah 40–55, this leads to identifying the servant with Israel, since the servant is identified by Jacob/Israel both in the majority of instances within these chapters, and at the first occurrence of the title "servant." Identifying the servant with a prophet necessitates a different original context of Isaiah 53. Commonly, this different context is suggested to be the four servant songs. Answering the question of the original context of Isaiah 53 requires a detailed study of both language, structure, and content of Isaiah 40–55, which could not be undertaken here. Nevertheless, the arguments for excising the servant songs do not seem to be compelling. Thus, it seems reasonable to assume that Isaiah 40–55 is the context of Isaiah 53. Consequently, the servant is assumed to be Israel, in particular, the exiled community.

However, the study of the text of Isaiah 53 shows that the rhetoric device of ambiguity is employed frequently. The frequency of the rhetoric device of ambiguity suggests that it is employed intentionally, in order to create an ambiguous text. Specifically, verse 11aγ has several meanings which seem to contradict each other, with some stating that the servant is in need of reconciliation, while others state that the servant is righteous, and thus not in need of reconciliation. Since the ambiguity seems to be intended, this suggests that the identity of the servant is ambiguous. On the one hand, the servant Israel, depicted as sinful in Isaiah 42:24, 43:22–28, 44, 21f, and 50:1, in need of reconciliation with God. On the other hand, a righteous servant, who reconciles the many. The righteous servant

is indicated to be a member of the exiled community, and at the same time, their ally.

The structure of Isaiah 53 suggests that the many comprise the many nations, the we-group, and the addressee of the poem. The we-group uses a designation commonly used for Israel. This indicates that the we-group is Israel. Thus, Israel is comprised by the many who are reconciled by the servant. This is substantiated by the we-group stating that they are reconciled by the servant. As a consequence, the addressee of the poem must be every nation.

Isaiah 53 is also ambiguous concerning how peace with God is achieved. It states both reconciliation by chastisement and insight, and by transfer of the burden of sin, and clearly alludes to various means of cultic expiation of sin.

Concerning chastisement and insight, Isaiah 53 depicts the servant Israel as suffering in exile. This suffering is "God's chastisement for our peace" (Is $53:5b\alpha$), fostering insight into the conduct required by God, and adhering to it. By adhering to it, the servant is righteous and proves himself righteous to the nations, as is expressed by one meaning of Is $11a\beta\gamma$, "By his knowledge, the righteous one, my servant, will prove himself to be righteous to the many."

However, a main concern in Isaiah 53 and Isaiah 40–55 is Israel's sin. Conducting a righteous life does not deal with the consequence of sin. Sin has to be atoned for, and if it is not atoned for, the consequences of sin must be endured.

Concerning transfer of the burden of sin, Isaiah 53 states that the we-group's and the many's burden of sin, that is, the guilt incurred by sin and the punishment for this guilt, is transferred to the righteous servant. His suffering is "God's punishment for our peace" (Is 53:5b α). "By his sweat, a righteous one will reconcile my servant before the many" (11a $\beta\gamma$). At the same time, "by his sweat, the righteous one, my servant, will reconcile the many" (11a $\beta\gamma$).

Such a transfer of the burden of sin amounts to vicarious suffering. The concept of vicarious suffering might be inspired by the concept of vicarious substitution. This concept is found in the Old Testament in relation to existential issues such as atonement and having sons. It is applied to vicarious punishment by Judah offering to substitute for Benjamin.

Concerning cultic expiation of sin, Isaiah 53 clearly alludes to both the Day of Atonement and to the liability offering. However, the servant is not decisively depicted as either one of them. Further, cultic atonement is not a theme in Isaiah 40–55, whereas God's judgment and punishment is a common theme in these chapters. Thus, Isaiah 53 seems to regard reconciliation by forensic transfer of the burden of sin rather than by cultic atonement. However, the allusions to cultic atonement suggest that the forensic reconciliation of transfer of the burden of sin happens within a framework of cultic atonement, and thus is atoning. The burden of sin is not only transferred, but by this transfer, the others' sin is completely annihilated. The others are completely reconciled.

Isaiah 53 presents two servants, Israel and a righteous servant, by frequent use of the rhetoric device of ambiguity. By the same device, Isaiah 53 presents reconciliation by chastisement, by transfer of the burden of sin, and within a cultic framework. As shown for Is 49:5f, the servant is presented ambiguously as Israel and a servant distinct from Israel also in these verses. Studying the motivation for the ambiguous way of presentation in Isaiah 53 therefore most likely requires a study of ambiguity in Isaiah 40–55, which could not be undertaken here.

This thesis has emphasized the many ambiguities of Isaiah 53. These pose an impossible challenge to any attempt to translate the text without listing all potential meanings of the ambiguous expressions. The common choice of translation can in some cases be doubted, for example for verse 11aγ. Of the many possible translations, NRSV chooses "the righteous one, my servant, shall make many righteous." However, if only one translation of 11aγ has to be chosen, then a more adequate choice might be "a righteous one will make my servant righteous before the many." Both translations conform with the parallel verse 11b, "and their sins' burden he will carry." The translation "a righteous one with make my servant righteous" has the advantage of being the straight-forward translation of the Hebrew sentence, and it is the translation which conforms best with the context of Isaiah 40–55, where the servant is identified with Israel. The translation chosen by NRSV makes Isaiah 53 tend strongly towards an interpretation of the servant as a righteous individual.

The thesis has also taken an uncommon point of view on some of the textcritical issues of Isaiah 53. Probably the most controversial of these is that יְרָאֶה in verse 11aa should not be supplied with an object, whereas most modern bible translations and the commentaries by Hermisson (2017b) and Berges (2015) add "light" as object to it. Against this addition, it can be argued that יְרָאֶה together with other expressions in Isaiah 53 alludes to Is 41:20, where אור ווי שׁבָּע בּרַשָּה is used without direct object. It can further be argued that the references in Isaiah 53 in general are varied. The variation of a verb being using in its transitive and intransitive form is employed also for בּגע הַּרְשָהוֹ here means "he will understand" and is used intransitively.

Bibliography

- Abegg, Martin G., et al. 1999. The Dead Sea Scrolls Bible. San Francisco: Harper. ISBN: 0060600632.
- Anderson, G. W., et al., editors. 1982. Theologisches Wörterbuch zum Alten Testament. Volume III. Stuttgart: Kohlhammer. ISBN: 3170076426.
- ———, editors. 1989. Theologisches Wörterbuch zum Alten Testament. Volume VI. Stuttgart: Kohlhammer. ISBN: 3170099345.
- ———, editors. 1993. *Theologisches Wörterbuch zum Alten Testament*. Volume VII. Stuttgart: Kohlhammer. ISBN: 3170126679.
- Balz, Horst, et al., editors. 2001. Theologische Realenzyklopädie: Bd. 32: Spurgeon Taylor. Berlin: de Gruyter.
- Bellinger, W. H., and William R. Farmer, editors. 1998. Jesus and the suffering servant: Isaiah 53 and Christian origins. Harrisburg, Pennsylvania: Trinity Press. ISBN: 156338230X.
- Berges, Ulrich. 2008. *Jesaja 40–48*. Edited by Erich Zenger et al. Freiburg im Breisgau: Herder. ISBN: 9783451268366.
- ——. 2015. Jesaja 49–54. Edited by Erich Zenger et al. Freiburg im Breisgau: Herder. ISBN: 9783451268373.
- Bergman, J., and G.J. Botterweck. 1982. «יֻרָּעֵּ» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., III:688–512. Stuttgart: Kohlhammer. ISBN: 3170076426.
- Bons, Eberhard, et al. 2011. Septuaginta Deutsch: Erläuterungen und Kommentare zum griechischen Alten Testament. Volume II Psalmen bis Daniel. Stuttgart: Deutsche Bibelgesellschaft. ISBN: 9783438051431.
- Branson, R.D. 1982. «יְסֵר» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., III:688–697. Stuttgart: Kohlhammer. ISBN: 3170076426.

- Childs, Brevard S. 2001. *Isaiah*. The Old Testament library. Louisville, Ky: Westminster John Knox Press. ISBN: 0664221432.
- Cover, Robin C. 1992. «Sin, Sinners». In *The Anchor Bible dictionary*, edited by David Noel Freedman, 6:31–40. New York: Doubleday. ISBN: 038526190X.
- Duhm, Bernhard. 1892. Das Buch Jesaia. 5. Auflage 1968. Göttingen: Vandenhoeck & Ruprecht.
- Fuhs, H.-F. 1993. «רָאָה» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VII:225–266. Stuttgart: Kohlhammer. ISBN: 3170126679.
- Gerleman, Gillis. 1976. «שלם šlm genug haben». In Theologisches Handwörterbuch zum Alten Testament: zwei Bände, edited by Ernst Jenni and Claus Westermann, II:919–935. München: Kaiser.
- . 1980. Studien zur alttestamentlichen Theologie. Volume 1978. Franz Delitzsch-Vorlesungen. Neue Folge. Heidelberg: Lambert Schneider.
- Gerstenberger, E. S. 1989. «שְנָה». In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VI:247–270. Stuttgart: Kohlhammer. ISBN: 3170099345.
- Goldingay, John, and David Payne. 2006a. The International critical commentary on the Holy Scriptures of the Old and New Testaments: Isaiah 40–55. Volume 1. London: T&T Clark. ISBN: 9780567173522.
- ——. 2006b. The International critical commentary on the Holy Scriptures of the Old and New Testaments: Isaiah 40–55. Volume 2. London: T&T Clark. ISBN: 9780567020000.
- Hägglund, Fredrik. 2008. Isaiah 53 in the Light of Homecoming after Exile. Volume 31. Forschungen zum Alten Testament, 2. Reihe. Tübingen: Mohr Siebeck. ISBN: 9783161497735.
- Hamp, V. 1982. «מְּבֶּשְ» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., III:302–306. Stuttgart: Kohlhammer. ISBN: 3170076426.
- Hermisson, Hans-Jürgen. 1987–2014. *Deuterojesaja*. Biblischer Kommentar Altes Testament XI. Neukirchen-Vluyn: Neukirchener Verlagsgesellschaft mbH.
- . 2003. Deuterojesaja. Volume 2. Biblischer Kommentar Altes Testament XI. Neukirchen-Vluyn: Neukirchener Verlagsgesellschaft mbH. ISBN: 3788712589.
- ———. 2017a. «Deuterojesaja». In *Das Wissenschaftliche Bibellexikon im Internet*. http://www.bibelwissenschaft.de/de/stichwort/16341/.
- . 2017b. *Deuterojesaja*. Volume 3. Biblischer Kommentar Altes Testament XI. Neukirchen-Vluyn: Neukirchener Verlagsgesellschaft mbH. ISBN: 3788730420.

- Hooker, Morna D. 1998. «Did the Use of Isaiah 53 to Interpret His Mission Begin with Jesus?» In *Jesus and the suffering servant: Isaiah 53 and Christian origins*, edited by W. H. Bellinger and William R. Farmer, 88–103. Harrisburg, Pennsylvania: Trinity Press. ISBN: 156338230X.
- Janowski, Bernd. 1982. Sühne als Heilsgeschehen: Studien zur Sühnetheologie der Priesterschrift und zur Wurzel KPR im Alten Orient und im Alten Testament. Volume 55.
 Wissenschaftliche Monographien zum Alten und Neuen Testament. Neukirchen-Vluyn:
 Neukirchener Verlag. ISBN: 3788706635.
- . 1993. «Er trug unsere Sünden: Jesaja 53 und die Dramatik der Stellvertretung: Hans Walter Wolff zum 17.12.1992». Zeitschrift für Theologie und Kirche 90 (1): 1–24. ISSN: 00443549.
- ——. 1997. Stellvertretung: alttestamentliche Studien zu einem theologischen Grundbegriff. Volume 165. Stuttgarter Bibelstudien. Stuttgart: Verlag Katholisches Bibelwerk. ISBN: 3460046511.
- ———. 2009. Ecce homo: Stellvertretung und Lebenshingabe als Themen Biblischer Theologie. Volume 84. Biblisch-Theologische Studien. Neukirchen-Vluyn: Neukirchener Verlag.
- Jenni, Ernst, and Claus Westermann, editors. 1971–1976. Theologisches Handwörterbuch zum Alten Testament: zwei Bände. München: Kaiser.
- Joachimsen, Kristin. 2011. *Identities in Transition: The Pursuit of Isa. 52:13–53:12.* Volume 142. Supplements to Vetus Testamentum. Leiden: Brill. ISBN: 9789004201064.
- Johnson, B. 1989. «צְּדַק» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VI:898–924. Stuttgart: Kohlhammer. ISBN: 3170099345.
- Joüon, Paul, and Takamitsu Muraoka. 1991. A grammar of Biblical Hebrew: Vol. 2: Part three: Syntax; Paradigms and indices. Roma: Editrice Pontificio Istituto Biblico. ISBN: 8876535950.
- Kellermann, D. 1973. «בּאָשֶׁה» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., I:463–472. Stuttgart: Kohlhammer. ISBN: 3170012096.
- Knierim, Rolf. 1965. Die Hauptbegriffe für Sünde im Alten Testament. Gütersloh: Gütersloher Verlagshaus.
- ——. 1971. «שְּׁשָׁה 'āṣām Schuldverpflichtung». In Theologisches Handwörterbuch zum Alten Testament: zwei Bände, edited by Ernst Jenni and Claus Westermann, I:251–257. München: Kaiser.

- Knierim, Rolf. 2001. «Sünde II. Altes Testament». In *Theologische Realenzyklopädie: Bd.* 32: Spurgeon Taylor, edited by Horst Balz et al., 365–372. Berlin: de Gruyter.
- Koch, Klaus. 1989. «קֶבֶּר» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VI:1149–1156. Stuttgart: Kohlhammer. ISBN: 3170099345.
- Koenen, K. 1993. «שָׁבֶל» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VII:781–795. Stuttgart: Kohlhammer. ISBN: 3170126679.
- Koole, Jan Leunis. 1998. Historical commentary on the Old Testament: Part 3 Vol. 2: Isaiah 49–55. Volume Part 3 Vol. 2. Kampen: KOK Publ. House. ISBN: 9042906790.
- Körting, Corinna. 1999. Der Schall des Schofar: Israels Feste im Herbst. Volume 285. Beihefte zur Zeitschrift für die alttestamentliche Wissenschaft. Berlin/Boston: De Gruyter. ISBN: 3-11-016636-4.
- Lindström, Fredrik. 1994. Suffering and Sin: Interpretations of Illness in the Individual Complaint Psalms. Volume 37. Coniectanea Biblica. Stockholm: Almqvist & Wiksell. ISBN: 9122015809.
- Maiberger, P. 1989. «שָּנֶש» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VI:501–508. Stuttgart: Kohlhammer. ISBN: 3170099345.
- Mettinger, Tryggve N. D. 1983. A farewell to the servant songs: a critical examination of an exegetical axiom. Volume 1982–1983:3. Scripta minora. Lund: CWK Gleerup. ISBN: 9140048721.
- Miller, Patrick D. 2000. The Religion of Ancient Israel. Louisville, Kentucky: Westminster John Knox Press. ISBN: 066423237X.
- North, Christopher R. 1956. The suffering servant in Deutero-Isaiah: An historical and critical study. 2nd ed. London: Oxford University Press.
- ——. 1964. The second Isaiah: introduction, translation and commentary to chapters XL-LV. Oxford: Clarendon. ISBN: 0198261543.
- Orlinsky, Harry M. 1977. «The so-called "Servant of the Lord" and "Suffering Servant" in Second Isaiah». In *Studies on the second part of the Book of Isaiah*, volume 14. Supplements to Vetus Testamentum. Leiden: Brill. ISBN: 9004054375.
- Oswalt, John N. 1997. The New International Commentary on the Old Testament: The Book of Isaiah Chapters 40–66. Grand Rapids, Michigan: William B. Eerdmans Publishing Company. ISBN: 0802825346.
- von Rad, Gerhard. 1967. Das Alte Testament Deutsch: Neues Göttinger Bibelwerk: Das erste Buch Mose: Genesis. 8. Auflage. Volume 2/4. Göttingen: Vandenhoeck &; Ruprecht.

- Sæbø, Magne. 1971. «יסר» jsr züchtigen». In Theologisches Handwörterbuch zum Alten Testament: zwei Bände, edited by Ernst Jenni and Claus Westermann, I:738–742. München: Kaiser.
- Sapp, David A. 1998. «The LCC, 1QIsa, and MT Versions of Isaiah 53 and the Christian Doctrine of Atonement». In *Jesus and the suffering servant: Isaiah 53 and Christian origins*, edited by W. H. Bellinger and William R. Farmer, 170–192. Harrisburg, Pennsylvania: Trinity Press. ISBN: 156338230X.
- Spieckermann, Hermann. 2001. «Stellvertretung II. Altes Testament». In *Theologische Realenzyklopädie: Bd. 32: Spurgeon Taylor*, edited by Horst Balz et al., 135–137. Berlin: de Gruyter.
- ———. 2004. «The Conception and Prehistory of the Idea of Vicairous Suffering in the Old Testament». In *The suffering servant: Isaiah 53 in Jewish and Christian sources*, edited by Bernd Janowski and Peter Stuhlmacher, 1–15. Der leidende Gottesknecht. Grand Rapids, Michigan: William B. Eerdmans Publishing Company. ISBN: 080280845X.
- Steck, Odil Hannes. 1984. «Aspekte des Gottesknechts in Deuterojesajas "Ebed-Jahwe-Liedern"». Zeitschrift für die Alttestamentliche Wissenschaft 96 (3): 372–390. doi:10. 1515/zatw.1984.96.3.372.
- . 1985. «Aspekte des Gottesknechts in Jes 52,13-53,12». Zeitschrift für die Alttestamentliche Wissenschaft 97 (1): 36-58. doi:10.1515/zatw.1985.97.1.36.
- Sweeney, Marvin A. 2016. *Isaiah 40–66*. Volume Vol. 17. Grand Rapids, Michigan: William B. Eerdmans Publishing Company. ISBN: 9780802866073.
- VanderKam, James, and Peter Flint. 2002. The Meaning of the Dead Sea scrolls: Their Significance for Understanding the Bible, Judaism, Jesus, and Christianity. New York: HarperCollins. ISBN: 006068464X.
- Vanoni, G. 1993. «שַּׁים» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VII:761–781. Stuttgart: Kohlhammer. ISBN: 3170126679.
- Waschke, E.-J. 1989. «צֹאַן» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VI:858–868. Stuttgart: Kohlhammer. ISBN: 3170099345.
- ——. 1993. «שֶׂה» In Theologisches Wörterbuch zum Alten Testament, edited by G. W. Anderson et al., VII:718–721. Stuttgart: Kohlhammer. ISBN: 3170126679.
- Whybray, R. N. 1978. Thanksgiving for a Liberated Prophet: An Interpretation of Isaiah Chapter 53. Reprinted with corrections, 1985. Volume 4. Journal for the Study of the Old Testament. Sheffield: JSOT. ISBN: 0905774043.

- Zenger, Erich, et al., editors. 1999. Herders Theologischer Kommentar zum Alten Testament. Freiburg im Breisgau: Herder.
- Zimmerli, Walther. 1969. «Zur Vorgeschichte von Jes. LIII». In *Congress Volume Rome* 1968, edited by G.W. Anderson et al., 62:236–244. Supplements to Vetus Testamentum. Leiden: Brill.